



Article Urbanization Heat Flux Modeling Confirms It Is a Likely Cause of Significant Global Warming: Urbanization Mitigation Requirements

Alec Feinberg 🕩



Citation: Feinberg, A. Urbanization Heat Flux Modeling Confirms It Is a Likely Cause of Significant Global Warming: Urbanization Mitigation Requirements. *Land* **2023**, *12*, 1222. https://doi.org/10.3390/ land12061222

Academic Editors: Lawrence Lau, Rajnish Kaur Calay and Isaac Yu Fat Lun

Received: 20 March 2023 Revised: 25 April 2023 Accepted: 2 May 2023 Published: 12 June 2023



Copyright: © 2023 by the author. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). DfRSoft Research, Physics Department, Northeastern University, Boston, MA 02115, USA; dfrsoft@gmail.com

Abstract: Recent ground-based measurements find the magnitude of the urbanization effect on the global average annual mean surface air temperature corresponds to an urbanization contribution of 12.7%. It is important to provide modeling to help understand these results as there are conflicting concerns. This study models the global warming contribution that urbanization heat fluxes (UHF) can make due to anthropogenic heat release (AHR), and solar heating of impermeable surface areas (ISAs), with additional secondary effects. Results help explain and support ground-based observations. Climate models typically omit anthropogenic heat release (AHR) as warming estimates are below 1%. In agreement, the baseline assessment in this paper has similar findings. However, in this study, the methods of climate amplification estimates (MCAE) with data-aided physics-based amplification models are used. When the MCAE are applied at the global and microclimate levels that consider greenhouse gases (GHGs), feedback, and other secondary effects; the results show that AHR fluxes can amplify, increasing to have an estimated global warming (GW) influence of 6.5% from 1950 to 2022 yielding a 0.9% decade⁻¹ increase. This increasing rate due to energy consumption is found as anticipated to be reasonably correlated to the increasing population growth rate over this time. Furthermore, using the MCAEs, this paper studies heat fluxes assessment due to solar heating of unshaded impermeable surfaces including likely secondary amplification effects. Impermeable surface areas (ISAs) such as asphalt roads, roofs, and building sides have been reported with high land surface contact temperatures (LSCTs) relative to non-ISAs and significantly found to contribute to urbanization warming. Results indicate that high-temperature unshaded impermeable surfaces (including building sides) are estimated to average around 10-11 °C above the earth's ambient temperature of 14.5 °C (showing albedo ISA estimates between 0.133 and 0.115 respectively); the ISA heat fluxes with secondary effects are estimated to have about a GW influence of 6.5%. This is broken down with average contributions of 4.0% from urban ISAs and 2.5% from rural ISA heat fluxes. Asphalt road ISA heat fluxes are estimated to have about a 1.1% global warming influence. Then the total UHF effect from ISAs and AHR with secondary effects is assessed in modeling to yield a combined average GW influence of 13% helping to confirm ground-based measurement results. Several key adjustment values were used for shading, cloud coverage, and rural-to-urban ISA ratios. Microclimate GHGs and related water vapor feedback (WVF) were assessed to increase urban warming by about 50%. As well an assessment of water vapor and radiation increases from UHF is provided. This study also shows the need to incorporate urbanization heat fluxes with secondary effects into climate models and indicates the necessity for Paris Agreement urban heat flux mitigation goals. Results also found that given average climate conditions, it is possible to mitigate much of the UHI effect with an albedo increase of 0.1 that is anticipated to lower the average impermeable surface temperatures by about 9 °C. Studies show this can be accomplished with cost-effective cool roads and roofs. Although roads are only estimated to occupy 14% of ISAs, changing roads from asphalt to concrete-type surfaces would improve reflectivity by about a factor of 5 and is estimated to mitigate about 5.5% of global warming. Unfortunately, the current overuse of black asphalt on pavements and roofs is highly dangerous to our environment causing UHI increases in heatwaves, excessive temperatures, and global warming issues and should be banned. Asphalt usage also reduces opportunities for solar geoengineering of urbanization.

Keywords: urbanization global warming; anthropogenic heat release; UHI global warming; microclimate modeling; UHI mitigation; solar heating of impermeable surfaces

1. Introduction

Recent ground-based measurements by Zhang et al.'s [1] found, "The magnitude of the urbanization effect on global land averaged annual mean surface air temperature change over 1951–2018 in this study is $0.03 \,^{\circ}$ C decade⁻¹ and the corresponding urbanization contribution is 12.7%." Part of the goal of this study is to help explain through microclimate data-aided physics-based and estimation modeling, the feasibility of their observation and provide root cause assessment. This paper is able to demonstrate modeling validation and help in our understanding of the root causes due to UHFs. The recent comprehensive work of Zhang et al. [1], the only worldwide study of its kind, has not been recognized in climate models and the IPCC AR6 report; the IPCC in AR6 [2] states that urbanization has a "negligible impact on global annual mean surface-air warming (very high confidence)." Such statements show the difficulty of understanding root causes and Zhang et al.'s [1] findings.

One may begin to understand Zhang et al.'s [1] results by realizing that population growth, energy consumption, and its CO_2 byproduct are highly correlated to global warming [3]. Although urbanization occupies only a small area of the earth, currently, 55% of the world's population lives in cities [4]. IEA [5] and Yang et al. [6] estimated that 75% of energy is consumed within cities. As well, automotive activity and fossil fuel usage generate increases in CO_2 in urban areas where Koerner and Klopatek [7] estimated that 80% of CO_2 originates. We also know that UHIs have high concentrations of heat fluxes that have been observed to exceed 400 Wm⁻² in city centers [8]. This increases the probability for local GHG pollution and humidity exasperation effects to interact with heat fluxes amplifying heat higher than the standard atmosphere [2,9].

In general, solar heating is the primary source of warming the earth while greenhouse gases (GHGs) are an important secondary re-radiation effect. Therefore, additional UHFs are concerning as they increase primary heating and can be amplified in the microclimate and the background global climate by secondary effects. GHGs and related water vapor feedback (WVF) and other secondary effects are important parts of climate science and need to be included in modeling.

In this study, there is also a focus on providing physics-based data-aided UHI microclimate and background global climate modeling of GHGs and related WVF and other amplification effects. This modeling can help explain how urbanization can contribute to global warming. The other key microclimate UHI effects are atmospheric aerosol pollutants, loss of evapotranspiration cooling, temperature inversion reduced cooling, and so forth.

This paper also studies anthropogenic heat release (AHR) which is understood to be a key contributor to the urban heat island. As well, this paper focuses on solar heating of ISAs such as asphalt pavements, rooftops, and the large solar area of building sides, as the other main contributor to the UHI. Therefore, in this portion of the study, assessments are limited to land cover from ISAs only which excludes areas such as farmlands. UHI heat fluxes are in the form of sensible and latent heat to the urban canopy and AHR is created mainly through heating and cooling energy consumption, running appliances, transportation, and industrial processes, which convert energy into anthropogenic heat release [10–14].

Climate models typically do not incorporate UHFs as initial assessments (without secondary effects) show only a small contribution (~2%, see for example Section 2.5) would result in global warming [2,6,14,15]. This leaves a gap in climate modeling with possible errors if these are key influences on global warming, as estimates indicate in this study and by Zhang et al. [1] measurements.

At the microclimate level, the 'energy in' for solar heating of ISA can be estimated with both the land surface contact temperatures (LSCTs) and the albedo effect. To make assessments, LSCT estimates are made in this paper relative to the global ambient temperature for unshaded ISAs. High LSCT is something we all experience in walking on unshaded dark pavements. It provides a helpful additional practical metric. These heat fluxes in the UHI environment can amplify, and modeling helps in assessments. For example, Zhao et al. [16] found that cities in humid compared to dry environments are 3.3 K hotter (see Section 2.2). Their data has been modeled as UHI water vapor feedback (WVF) recently [17] and related amplification estimates are provided in Section 3.1.4.

Many land surface temperature studies and ground measurement station observations are taken from mean surface air temperatures at a height of about 5–6.5 feet [18] which are significantly lower in temperatures than LSCT from solar heating of asphalt roads, for example, that can often be higher than 20 °C above ambient. Zhang et al.'s [1] ground station measurements were taken at about this height. Given that they found a significant urbanization effect at about 5–6.5 feet, it is likely that LSCT heat fluxes from solar heating of ISAs are making substantial contributions in their observations compared to AHR.

A recent satellite study out of China by Guo et al. [19] found in 10 of 11 cities, albedo estimates appear to average about 0.16. Guo et al. [19] deduced a substantial cooling effect from the temporal history of the "relationship between land surface albedo and land cover in cities". These are land-cover types of measurements with the relationship between surface albedo and urban canopy not fully known. However, this leaves AHR and their cooling losses as the main root causes for the UHI effects in these megacities according to their results. It is interesting to note that the results in this study on heat fluxes at the global level from solar heating of ISAs, an albedo of about 0.16, found that LSCT would average about 8.5 °C above ambient for unshaded areas and impact about 2.7% of global warming.

It is important to know how comparisons are made in studies. Cities have higher microclimate local amplification compared to rural areas so equating albedo assessments is difficult. It is pointed out in this paper, that it can be helpful to incorporate amplification effects when appropriate and also consider making additional comparisons relative to ambient conditions or the natural reflectivity of land which is about 25% [20] or the earth (~30%) as this provides valuable information. Using such comparisons, for example, Guo et al. [19] would likely deduce albedo UHI warming rather than cooling effects (see Section 4.1). Further, it implies that albedo UHI goals should be higher to offset warming from solar heating of ISAs, AHR, the microclimate GHG amplification, and the cooling losses that can occur in cities.

This study also provides modeling for both AHR and ISAs which includes building effects. The impact of solar heating due to building sides and related solar canyon effects are not well understood in GW albedo land satellite studies (see Section 4.2) along with microclimate amplification effects. Often climatologists favor satellite compared to full ground-based direct measurements and these can appear in conflict with GW urbanization assessments [21] (see Section 4). In general, there is a long history of conflicts among authors on the effects of urbanization warming that modeling in this study may help resolve.

Document Smart Diagram

To aid the reader in this two-part study, a Table 1 smart flow chart is provided below. Each part of the study is laid out in the table.

Urbanization Heat Flux Assessment									
Urban Flux Type	Amplificat	ion Models	Model Verifications	Flux Type Results	Results				
Part 1: 3.1 AHR GW Baseline Data Sections 2.4, 2.5 and 3.1.9 → Part II: 3.2 ISA GW Baseline Est. Section 3.2.1	Tables 2 and Aerosol Etc Global backgrou UHI GHG: Section 3.1. UHI WVF Sections 2 Dry & Wet Mixed S Footprint Sections 2. Rural Section	d 3, Figure 1 . Section 2.3 .nd Section 3.1.1 3, Appendix D Model 1 2.2 and 3.1.4 Model 2 ection 3.1.5 Model 3 1.1 and 3.1.6 Model 4 3.1.7 Model 5 \rightarrow Figure 1	Wet & Dry Model 3 Footprint Model 4 Section $3.1.10$ Tables $3-5$ \rightarrow	AHR 6.5% GW: Section 3.1.8, Equation (14) \rightarrow ISA 6.5% GW: Section 3.2.1 Table 6 \rightarrow	AHR+ISA GW 13% GW Section 3.2.2 Table 7				
Mitigation of UHI Urban SG require Helpful Equations:	² Sections 3.3 and 4.1 ments, Section 3.3.2 Appendices A and B	Mitigation & He GW Breakdown of Ener Global background fe Sections 2, 2.5, 3. Appendices	elpful Information rgy Flux, Appendix F eedback, Tables 2–4 1.1 and 3.2.1.3 , s C and F	UHI WVF, Appe Irradiance, Se Satellite Issues, Secti	endices E and F ection 3.2.1.1 on 4.2, Appendix G				

Table 1. Smart Flow Chart.

2. Key Data and Methods

The key data used in modeling is overviewed in Table 2 with their applicable sections.

Table 2. Key Data l	Required in	Modeling.
---------------------	-------------	-----------

Data Type	Values	Description	Reference(s)	Applicable Sections
Fossil Fuel	176,431 TWh	Consumption in 2021	[22]	Section 2.4
UHI WVF	$3.4 \ Wm^{-2} \ K^{-1}$	UHI WVF in humid climates	[17]	Sections 2.2 and 3.1.4
Global WVF	$1.6 \mathrm{Wm^{-2} \ K^{-1}}$	Global WVF	[23]	Section 3.1.4
Global Warming	5.1 Wm ^{−2} , 0.95 °C	From 1950-2019	[9]	Sections 2.5, 2.1.1 and 3.2.1.2
Forcing	$2.38 \ Wm^{-2}$	From 1950-2019	[9]	Sections 2.3, 3.1.8 and 3.2.1.2, Appendix F
UHI ΔT Increase	3.3 K	UHIs ΔT increase in wet cities	[16]	Sections 2.2 and 3.1.4
Cloud Percent	47%	Irradiance through clouds	[24]	Section 3.2.1.1
Unshaded% ISAs	65%, 75%	Urban & rural unshaded %ISAs	Estimated, solar canyon	Section 3.2.1.1
Global Feedback	2.15	WVF Amplification Factor	[9]	Sections 2.2, 2.3, 3.1.8 and 3.2.1.2
Global GHG Factor	1.62	GHG Amplification with 62% re-radiation	[9,24]	Sections 3.1.1, 3.1.8 and 3.2.1.2
UHI Footprint	3.2	UHI Amplification model	[25], Section 3.1.10.1	Sections 3.1.1, 3.1.6, 3.1.10.1 and 3.2.1.2
% Urban Climate	67% vs. 33%	Cities in wet versus dry climates	[26]	Section 3.1.5
ISA	1.29 M km ²	0.255% of the earth	[27]	Section 3.1.9
ISA Roads	14% of ISA	Road % of ISA	[28]	Section 3.2.2
ISA% Urban	33%, 40%, 50%	Urban vs. Rural %ISA	[28]	Sections 3.1.9, 3.2.1.1, 3.2.1.2 and 3.2.2

In GHG re-radiation modeling, at the global level secondary effects can be applied to energy terms. For example, GHGs increase the warming of the earth from about -18.6 °C to 13.9 °C. In the absence of global warming, a simple re-radiation model illustrates how this amplification can be applied [9]

$$P_{Total} = P_{\alpha} + \overline{f} P_{\alpha} = P_{\alpha} \left(1 + \overline{f} \right) = A_{GHG} P_{\alpha} \tag{1}$$

Here P_{α} is the solar 'Energy In' equivalent to the 'Energy out' (in Wm⁻²), \overline{f} is the average GHG re-radiation value of about 62% [9]. In this paper, we can consider the secondary GHG re-radiation term $(1 + \overline{f})$ as a GHG amplification factor (A_{GHG}), shown in Table 3 to equate to 1.62. As well, amplification due to feedback effects can be applied similarly [9]. In Table 3, an overview of secondary factors for the MCAE is presented and applied to urbanization heat flux estimates both globally and locally. The major local

secondary microclimate effect root cause from the table occurs in humid environments. Furthermore, studies suggest that local CO_2 GHGs can impact this UHI intensity in both dry and humid environments. In Sections 2.1–2.3 and 3.1.2–3.1.5, overviews and assessments are provided on the influence of these local urban secondary effects.

Secondary Effect	Amplification Factor	Reference							
	Global Amplification E	stimates							
Global re-radiation GHGs	1.62	[9,24]							
Global feedback (water-vapor and other	(>2)	[9,23,29,30]							
Global combined amplification effects	3.48 = 1.62 × 2.15								
Microclimate	e Amplification Estimates (humid vs. dry climates)							
Physics-based Modeling	1.2	Model 1: Section 3.1.3 dry climates							
Physics-based Modeling	2.55	Model 2: Section 3.1.4 humid climates							
Physics-based Modeling	2.2	Model 3: Section 3.1.5, mixed climates							
UHI Estimate	3.2	Model 4 (footprint): Section 3.1.6							
Rural Area Estimate	1.1	Model 5: Section 3.1.7							
Global ×	Global × Microclimate (Local) Amplification Estimates								
Global & UHI local combined amplification effects	$1.62\times2.15\times A_{Local}$	Sections 3.2.1 and 3.2.2, (Example in Equation (6))							

Table 3. Global and local urban secondary factors for the MCAE.

As an overview of the MCAE, in this paper, the general microclimate model mixed with global background climate amplification factors is as follows

$$P_{A-UHF} = \left(\left[A_U P_{ISA-U} + A_{Ru} P_{ISA-Ru} \right] + \left[A_U P_{AHR-U} + A_{Ru} P_{AHR-Ru} \right] \right)_{Micro} A_{GR} A_{GF}$$
(2)

In the equation, P_{A-UHF} is the total amplified urban heat flux. In the microclimate portion right-hand side of the equation A_U , A_{Ru} are the urban and rural amplification factors, P_{ISA-U} , P_{ISA-Ru} are the urban and rural solar heated ISA fluxes, P_{AHR-U} , P_{AHR-Ru} are the urban and rural AHR fluxes. The microclimate heat fluxes are released into the atmosphere as out-going longwave radiation and then can be amplified by the global background climate assessed by A_{GR} , and A_{GF} , the global background re-radiation and feedback amplification factors of 1.62 and 2.15, respectively in Table 3. Table 3, A_U is assessed by Model 3 and 4 separately and averaged while A_{Ru} is identified as Model 5. Stored sensible heat emissions from buildings with large heat capacities are typically released over the daily cycle (Section 3.2.1.1) and therefore intrinsically included in the P_{ISA} and P_{AHR} terms.

UHFs from ISAs are dominated by longwave radiation and sensible heat to the atmosphere while AHR has been broken down by some authors with a small portion resulting as latent heat. However, in climate modeling, it is often treated as consisting of longwave radiation and/or sensible heat [13,31] to the atmosphere. We assume the heat absorbed in the lower atmosphere will turn up into longwave radiation and treat it as other authors do in this type of climate modeling [32] as longwave radiation.

2.1. The Global Background, Microclimate, and the Global Warming Flux

2.1.1. Global Background Amplification

All UHFs are subject to increase due to the global background climate. There are two global climate amplifying effects. UHFs can be increased by the re-radiation of GHGs. About 62% re-radiation is anticipated yielding a 1.62-factor increase. UHFs can also be

increased in the global climate by feedback that is dominated by water vapor. This is a 2.15 factor. Both of these effects are discussed in Section 3.1.1 and explicitly shown in each calculation where applicable. When the background global climate is included in UHF estimates, global warming percentages due to urbanization effects are then estimated relative to 5.1 Wm⁻² which represents the radiative warming due to forcing and feedback that occurred from 1950 to 2019. This number is broken down in Appendix F and Equations (5)–(7) illustrate its use.

2.1.2. Estimating Microclimate UHI Amplification

Physics-based UHI microclimate modeling and estimations (See Figure 1) supported by data are provided in Section 3.1.3, Section 3.1.4, Sections 3.1.5–3.1.7. There are five models: Model 1 for dry climates, Model 2 for humid areas, Model 3 for mixed-weighted dry-humid climates, Model 4 is a footprint amplification type estimate added to Model 3, and Model 5 for rural GHG dry and wet microclimates.



Figure 1. Visualization of amplifications of main heat fluxes in UHIs (not to scale). Arrows A is the main upwelling (UW) from initial heat flux either due to ISA solar absorption and/or AHR. B is the downwelling from secondary WVF with GHGs, CO₂, and/or methane. Then UW-1 is the resulting total amplified heat fluxes from A and added B due to GHG re-radiation. On the right portion, the dashed line A, has added upwelling C component resulting from the solar canyon absorbed shortwave radiation. The other types of secondary downwelling or intensifying heat are D, due to possible temperature inversions (IV), E pollution (Pol), and F loss of cooling (LC) effects such as wind reduction, loss of evapotranspiration, etc. Then UW-2 is the resulting combined amplification magnitude (small dashed arrows) is often assessed with UHI footprint and dome measurements. The inner circle is the city limits (light gray), and the outer circle (dark gray) is the observed footprint from heat spreading beyond the city limits.

It is assumed in this study that UHIs dominate urban effects. The initial work on microclimate amplification by the author [25], assessed UHI amplification factors (A_{UHI}) based on UHI footprint [33] and dome estimates [34]. These are depicted in Figure 1. In this paper, modeling with data is used to help justify the prior footprint amplification value of 3.2 (Model 4) with the data assessment in Section 3.1.10.1. This value is also found to confirm Zhang et al.'s [1] findings yielding modeling consistency.

Zhou et al. [33] found an average footprint of 3.2 times the urban area. Feinberg [25] estimated that the overall urban areas growth from 1950 to 2019 increased also by a factor of 3.2 and dome areas increase by a factor of 8.4. The footprint and dome growth are indications of amplified heat flux that is observed to spread beyond the boundaries of the city itself both horizontally and vertically, respectively as depicted in Figure 1. As Figure 1 shows, heat fluxes amplify and increase in UHI microclimate for several reasons

7 of 34

including local water vapor feedback (WVF) or GHG re-radiation, solar canyon effect, heat inversions, pollution, and loss of cooling. These amplify heat spreading beyond the area of the city itself creating UHI footprints and dome effects. Zhou et al. [33] warn that "ignoring the footprint (FP) may underestimate the UHI intensity in most cases. Yang et al. [6] in a temporal study from 2003 to 2016 on 302 cities in China showed that day-night footprints are rapidly growing at an alarming rate of about 4.4% per year. This is an indication of a rapid spreading of 'heat pollution' (i.e., excess heat fluxes) in China and may also be occurring worldwide. It is an indication of the increasing local amplification issues as well. In this study, we provide a mixture of data-aided GHG physics-based amplification models and also use additional estimates for other effects which help support Zhang et al. [1] study.

2.2. UHI Amplification in Humid Environments

Many authors have reported more intense and frequent heatwave (HW) issues in humid compared to dry UHIs [35–43]. Authors found a strong amplification effect, for example, Russo et al. [38] found "the magnitude and apparent temperature peak of heatwaves, such as the ones observed in Chicago in 1995 and China in 2003, have been strongly amplified by humidity". In the very humid wet climate of Bangladesh, Dewan et al. [44] found, "Results indicated that annual surface UHI intensity was greater in the larger cities of Dhaka and Chittagong than in the smaller cities. Surface UHI intensity observed during the day was also greater than at night." Wang et al. [34] studied intensified humid heat events under global warming and found "Humid heat events show intensifications to dry heat events with higher frequency, duration, and intensity".

Empirical HW evidence exists for synergistic effects [37,40]. Liao et al. [40] in a study on HW exacerbation from UHIs in wet versus dry climates in China also showed a strong synergistic heatwave effect observed both in intensity, duration, and frequency. Liao [40] also noted that "in wet climates, the increasing trends of HWs in urban areas are greater than those in rural areas, suggesting a positive contribution of urbanization to HW trends".

Li and Zeid [37] studied the Baltimore-Washington metropolitan humid areas and found "synergies between urban heat islands and heatwaves. That is, not only do HWs increase the ambient temperatures, but they also intensify the difference between urban and rural temperatures. As a result, the added heat stress in cities will be even higher than the sum of the background urban heat island and the heatwave effect".

Zhao et al. [16] compared similarly constructed cities with twenty-four located in the humid southeastern United States to 15 cities in dry climates. They found an average ΔT increase of 3.3 K observed in daytime hours in humid climates with little differences in nighttime hours. This data is key in supporting Models 2, 3, and 4. Zhao et al. [16] attributed the humidity amplification effect to a reduction in sensible heat convection cooling efficiency. However, the observations by Zhao et al. [16] and other authors such as Dewan et al. [44] found the humidity effect occurs primarily in the daytime suggesting other mechanisms as well related to an atmospheric UHI humidity amplification mechanism. In a recent atmospheric study by Zhang et al. [45], urbanization atmospheric effects were observed to have a strong complex influence on extreme surface air temperature, where parameters studied were cloud cover, water vapor pressure, and sunlight duration. In their study, water vapor pressure was a key factor for humidity warming issues. Yang et al. [14] in east China (a humid environment in summer) found, "Overall, urbanization contributes to more than one-third of the increase of intensity of extreme heat events in the region, which is comparable to the contribution of greenhouse gases".

We suggested in prior work [17] and this paper that the UHI humidity effect is also likely a dual mechanism with GHG re-radiation and this can also relate to UHI convection cooling efficiency losses. Indeed, in daytime hours, solar heating fluxes are strongly increased in UHIs, and since warm air holds more water vapor in humid environments (see Section 3.3); this suggests that the ΔT increase (~3.3 K) in cities relative to rural environments in humid vs. dry climates, supports a likely GHG mechanism as well. Independent of the actual mechanism, the water vapor feedback equation is a function of the specific humidity and temperature changes. To quantify Zhao et al. [16] data, Feinberg [17] assessed it with a UHI temperature difference water vapor feedback model. The mathematical treatment found a UHI local feedback value of $3.4 \text{ Wm}^{-2} \text{ K}^{-1}$ [17] for cities in humid environments at 15 °C. This is about a factor of 2.1 higher than standard atmospheric global estimates (see Section 3.1.4, Equation (8)) but is much lower than the average UHI footprint and dome-expanded thermal area estimates discussed in Section 2.1. The temperature difference feedback model was verified by using it to reproduce global water vapor feedback estimates and is discussed in Section 3.1.4 with data verification in Section 3.1.10.

2.3. Urban Pollution of Aerosol, Haze, and other Similar Effects

Aerosols concentrations can reflect, and absorb solar radiation, and can reduce the amount of shortwave radiation reaching the ground producing a UHI cooling effect [46–48]. Guo et al. [19] observed a cooling effect in megacities in China. Aerosol reflectivity may have contributed to albedo increases.

However, Cao et al. [49] found that "the biogeochemical effect of urban aerosol or haze pollution is also a contributor to the UHI depending on particle size. Our results are based on satellite observations and urban climate model calculations. We find that a significant factor controlling the nighttime surface UHI across China is the urban-rural difference in the haze pollution level. The average haze contribution to the nighttime surface UHI is 0.7 ± 0.3 K (mean ± 1 s.e.) for semi-arid cities, which is stronger than that in the humid climate due to a stronger longwave radiative forcing of coarser aerosols".

Aerosols and haze pollution effects differ from city to city and are not easily modeled. These are perhaps best assessed as part of footprint and dome observations shown in Figure 1 (see Section 2.1 and Model 4).

2.4. AHR Data and Method

The total global energy production and consumption increase for coal, oil, natural gas, and other sources were calculated by Ritchie and Roser [22] up to the year 2021; this is shown in Figure 2 by the solid line. In this paper, it is assumed that 90% of this energy consumption ends up as heat (discussed below). A focus is on the time period relative to 1950 since this is suggested by the knee of the curves in Figure 2 and about the period of time covered in Zhang et al.'s [1] study. The annual AHR portion in 2021 is simply divided by the total area of the earth to obtain the global warming power per unit area and then compared to the value for global warming from 1950 to 2021. Then we considered both local and global secondary amplification effects in Table 3 based on prior studies [50] and modeling in Section 3.

We note in Figure 2, there appears a reasonable correlation between energy consumption shown by the solid line [22], and the global warming dashed line [51]. This correlation warrants further investigation [3]. It is also interesting that the slope of the energy curve is about 2196 TWh/year from 1975 to 2021, and in the last 5 to 10 years, the yearly amount has averaged 2133 TWh/year indicating the high rate of energy consumption increase per year has not significantly slowed.

Many authors approximate that 100% of the energy consumed eventually turns into heat. For example, Flanner [15] states, "Most energy used for the human economy is immediately dissipated as heat, either because of inefficient conversion processes or to serve the purpose of heating. Utilizing the second law of thermodynamics, it is assumed that all non-renewable primary energy consumption is dissipated thermally in Earth's atmosphere". According to the second law, there are no reversible processes so heat dissipation always occurs. According to the first law, energy use relates to some kind of loss of internal energy at the system's source supplying the energy which can go into work, heat, and sometimes part of the energy may be stored elsewhere. However, the work conducted relates to some type of kinetic energy change which eventually dissipates through frictional losses yielding more heat. We note that while many manufacturing processes dissipate heat, useful energy also creates long-lasting products such as buildings (with mgh potential energy), batteries, etc., which have some new stored internal energy. Other energy escapes to outer space without causing any heating similar to city night light energy. However, the actual energy consumed that turns readily into heat is not well established but is always estimated to be a high percentage (usually 100%) in climate models [14,15,52]. According to Ritchie and Roser [22], for example, a large portion of the energy consumed is in the northern areas, likely used primarily for direct heating. To be reasonably consistent with other authors and assess feasibility, we assume 90% of consumed energy eventually turns into heat.



Figure 2. Global energy consumption [22] in TWh (**left** axis solid line) compared to global warming trend [51] in °C (**right** axis dashed line).

2.5. AHR Baseline Estimates and Global Background Amplification

The total global energy consumption calculated by Ritchie and Roser [22] by 2021 has increased to 176,431 TWh annually (see Figure 2). Using a factor of 90% for an estimate of energy consumption converted to AHR, we find that annual power per year dissipated is

$$p_{AHR}(2021) = (176, 431 \times 10^{12} \text{ Wh}/8760 \text{ Hrs}) \times 90\% = 18.13 \times 10^{12} \text{ Watts}$$
 (3)

Here small p_{AHR} denotes the AHR power dissipated. Considering the area of the earth, then the total global warming (GW) energy per unit area denoted by capital P_{AHR} is

$$P_{AHR}(2021) = 18.13 \times 10^{12} \text{ Watts} / 510 \times 10^{12} \text{ m}^2 = 0.0356 \text{ Wm}^{-2}$$
 (4)

By 2021, the estimated GW since 1950 is about a 0.95 °C temperature rise which approximately corresponds to the time period in the Zhang et al. [1] study. Then the 0.95 °C temperature rise equates to about 5.1 Wm^{-2} (see for example Feinberg, [9], or one can use the Stefan-Boltzmann Law). This yields a baseline percentage of GW due to AHR yielding

$$P_{AHR}(2021) = 0.036 \text{ Wm}^{-2} / 5.1 \text{ Wm}^{-2} = 0.71\%$$
 (5)

Therefore, without considering any secondary effects, baseline data indicates about a 0.7% global warming influence which is comparable to what other authors have reported [14,15].

3. Results

We divide calculations into two sections for contributions from AHR (Part 1) and solar heating of ISAs (Part 2) for global warming estimates (see the smart diagram in the introduction). We start with AHR assessments as the calculations are simpler. Dataaided physics models and estimation assessments of microclimate amplification effects are also found in Part 1 (Sections 3.1.2–3.1.7). As well in Appendix F, we discuss UHF radiation forcing with water vapor increase estimates. In Section 3.3, UHF mitigation goals are presented.

3.1. Part 1: AHR Estimates

3.1.1. AHR Global Secondary Effects

Global background climate secondary estimates for greenhouse gases and feedback have been assessed in climate science IPCC models [24] and by the author [9]. Then, it is simplest to use well-established traditional climate science background global secondary estimates first. As indicated in Table 3, the potential global background average greenhouse gas re-radiation is 62% yielding a heat fluxes amplification factor of 1.62 in Equation (1) [9,24]. As well feedback (primarily due to water vapor) has the potential to more than double the heat flux forcing [23,29,30]. A 2.15 feedback amplification factor has been estimated by Feinberg [9]. These global secondary effects can increase AHR flux in Equation (4) warming to

$$P_{AHF-GS}(2021) = 0.036 \,\mathrm{Wm^{-2}} \times 1.62 \times 2.15 = 0.124 \,\mathrm{Wm^{-2}} \tag{6}$$

Here P_{AHR-GS} is the AHR with the background global secondary effects (in which we have relatively high confidence in global amplification estimates in climate science).

Similar to Equation (5), we can divide Equation (6) by 5.1 Wm^{-2} to obtain the percentage of GW due to AHR with global secondary effects to find

$$P_{AHF-GS}(2021) = 0.124 \,\mathrm{Wm^{-2}}/5.1 \,\mathrm{Wm^{-2}} = 2.43\%$$
 (7)

This yields the percentage of GW due to AHR when the global background climate amplification is applied.

Note the 5.1 Wm⁻² consists of forcing and feedback that is detailed in Appendix F for clarity. Therefore, urban estimates in this paper, as shown in Equation (6), are amplified by the global background re-radiation as well as the feedback factor 2.15. That is, feedback approximately doubles the urbanization forcing (similar to CO_2 energy fluxes). Therefore, as shown in Equation (6) urban forcing percentages are increased by the re-radiation factor 1.62 and the 2.15 global feedback effect. This means that percentages are relative to the 5.1 Wm⁻² as in Equation (7) and not relative to total forcing estimates which can be uncertain. In all calculations, this is explicitly shown including in the Tables (where applicable).

3.1.2. UHI Microclimate Amplification Models

Cities amplify heat. In general, any hotspot on the earth, such as Death Valley or UHIs may amplify or intensify solar heat. For example, a common reason that land intensifies heat is often due to loss of cooling (LC) from wind reduction or evapotranspiration loss (Figure 1F). This is common in UHIs. A key microclimate amplification mechanism considered in this study is related to UHI GHGs (Figure 1B) occurring in dry (Section 3.1.3) and humid climates (Sections 2.2 and 3.1.4). Other UHI amplification mechanisms relate to temperature inversions, haze/aerosol pollution, and loss of cooling (Figure 1D–F).

There are five models provided to assess microclimate amplification effects in Sections 3.1.3–3.1.7. Models are mainly data justified in these sections and Section 3.1.10.

3.1.3. UHI GHG Amplification Estimate in Dry Climates: Model 1

From Equations (1) and (4), we note that due to the high average AHR energy in UHIs (see Equation (14)), and increases in GHGs (mainly CO_2 in dry climates, see Appendix D) compared to rural areas, there is a higher probability for IR absorption by CO_2 molecules leading to increases in subsequent re-radiation events compared to the standard atmosphere. We note that due to its density, CO_2 is increased in the lower atmosphere. As well, automotive activity and fossil fuel usage generate increases in CO_2 in urban areas

where 80% of CO₂ is estimated to occur as discussed in Appendix D. This can amplify and increase heat fluxes higher than the standard atmosphere. Based on an overview of increased CO₂ data observations in cities in Appendix D, we take this increased GHG concentration (mainly due to CO₂) to average conservatively about 20%. This value is justified in Section 3.1.10 for surface temperature change due to CO₂ observations related to Table 4. In that section, data is cited on over 300 megacities in China during the lockdown by Liu et al. [23] when CO₂ was minimal reducing temperatures. Their data matched fairly well with the proposed Model 1. Here in Model 1 data in Appendix D leads to a 1.2 factor in GHG concentration increase in cities compared to the standard atmosphere. Recall, that exact estimates are not part of the goal, providing feasibility and root cause estimates aided by data are the main objectives of this study. Then a simplistic physics-based local amplification estimate can be made for cities relative to the standard background atmosphere, where in the microclimate we estimate an increase in atmospheric re-radiation amplification of

$$A_{U-Dry} = \frac{C_{GHG-U}}{\overline{C}_{GHG-S}} = 1.2 \qquad \text{Model 1}$$
(8)

Here $\overline{C}_{GHG-U}/\overline{C}_{GHG-S}$ is simply the ratio of the estimated increase of GHGs (primarily CO₂) in urban compared to the standard atmosphere taken as 1.2 and as stated, this value is justified in Section 3.1.10 by CO₂ data observations (Appendix D) with resulting temperature changes compared well to model predictions in Table 4. Model 1 can be used both in dry and humid atmospheres.

3.1.4. UHI Amplification Estimate in Humid Climates: Model 2

As discussed in Section 2.2, UHI temperatures are increased in humid versus dry climates relative to rural areas. At the background global atmospheric level, there are several estimates from different authors on the water vapor feedback (WVF) effect. For example, Gordon et al. [30] estimated 2.4 Wm⁻² K⁻¹, Dessler, et al. [29] found 2.2 Wm⁻² K⁻¹, and Liu et al. [23] estimated a value of 1.6 Wm⁻² K⁻¹. In Section 2.2, a discussion assessed that microclimate UHI amplification was perhaps best quantified mathematically with a recently provided temperature difference water vapor feedback model by the author [17] that could be applied to the mean ΔT observations from Zhao et al., [16] data in humid compared to dry UHI climates. When the WVF model was applied to Zhao et al. [16] temperature difference data for cities in humid environments that averaged 3.3 K hotter than dry environments, results found that in humid cities, the local water vapor feedback at 15 °C averaged 3.4 $Wm^{-2} K^{-1}$ (Feinberg, 2022b). This is perhaps a best method for assessing the observed UHI humidity amplification effect for making comparisons to the standard atmosphere. Furthermore, the temperature difference WVF model [17] was verified by reproducing the global results of Dessler et al. [29], Gordon et al. [30], and Liu et al. [23]. Liu et al.'s [23] value of 1.6 Wm⁻² K⁻¹ reproduced the most accurate results according to the temperature difference WVF model (and is also the most recent estimate). Then relative to the background climate average standard atmosphere (see Appendix C.2), the increased microclimate amplification estimate in humid UHIs is about

$$A_{U-WVF} = 3.4 \,\mathrm{Wm^{-2}K^{-1}}/1.6 \,\mathrm{Wm^{-2}K^{-1}} = 2.125 \tag{9}$$

This model, which is part of Model 4, is supported with alternative data in Section 3.1.10.1. Here A_{U-WVF} is then a ratio of WVF estimates between the local and standard atmosphere observed in UHI humid climates. In humid (denoted as Wet) UHI climates, Equation (8) also applies so that the total amplification is

$$A_{U-Wet} = A_{U-WVF} \times A_{U-Dry} = 2.55 \qquad \text{Model 2} \tag{10}$$

Note that this is below the UHI footprint estimate of 3.2 discussed in Section 2.1. The UHI area footprint amplification estimates incorporate many mechanisms such as loss of evapotranspiration cooling, temperature inversions, aerosols, and so forth. Therefore,

Model 2 is likely conservative and helps to demonstrate the feasibility of Zhang et al.'s [1] observations. We also provide support for this estimate in Section 3.1.10.1.

3.1.5. Dry and Humid UHI Mixed Climate Amplification Estimate: Model 3

To be able to apply microclimate UHI amplification estimates, a weighted mixed wet-dry climate assessment is helpful. This will be exemplified in the next section for AHR GW estimates. Cherlet et al. [26] break down the percentages of cities in humid versus dry climates. In terms of climate zones, the population distribution closely follows big cities that are estimated with 33% of the population living in drylands and 67% in non-dryland or humid zones. Interestingly, this closely follows the fact that 71% of the earth's composition consists of oceans.

Given the Cherlet et al. [26] breakdown for dry and humid climates, a mixed dryhumid amplification value due to Models 1 and 2, indicates

$$A_{UHI_mixed} = (0.33A_{U-Dry} + 0.67A_{U-Wet}) 1.05 = (0.32 \times 1.2 + 0.68 \times 2.55) 1.05 = 2.2 \text{ Model 3}$$
(11)

According to Cherlet et al. [26], the dry region estimate of 33% consists of 11% subhumid and 15% semiarid, conditions. This amounts to about a 5% increase correction in Equation (11) which is represented by the 1.05 factor. The details for this small 5% correction are provided in Appendix E.

3.1.6. UHI Footprint Amplification Estimate: Model 4

Model 4 is based on estimates for the heat flux increase requirements in Section 3.1.10.1 to support the data observations by Zhao et al. [16]. In that section, it is shown that to verify Zhao et al. [16] data observations that UHIs in wet climates are 3.3 K higher than in dry areas, an increase in amplification is required larger than the results of Model 3. This value also is found to help confirm Zhang et al.'s [1] findings adding to modeling self-consistency. This increase is also supported by prior studies [25] that take into account estimates for UHI footprints. The UHI footprint is an indication of amplified heat spreading beyond the city boundaries. A footprint type of amplification estimate, shown in Figure 1, may be conservative. As discussed in Section 2.2 and Figure 1, footprint estimates go as the city areas [33], and assessments show that area growth from 1950–2019, the period of interest, is about 3.2 times, and dome estimates are about a factor of 8.4 [25]. This 3.2 value also matched the footprint assessment by Zhou et al. [33]. Using the lower footprint estimate (Section 2.1) and the results in Section 3.1.10.1 to justify this increase, a footprint (FI) factor is applied to Model 3 of $\overline{A}_{\text{FI}} = 1.45$ in Equation (11) yielding the overall UHI footprint amplification ($A_{Footprint}$) estimate

$$A_{Footpring} = A_{UHI_mixed} \times A_{FI} = 2.2 \times 1.45 = 3.2$$
 UHI Footprint Model 4 (12)

The increase due to $A_{FI} = 1.45$ essentially takes into account Figure 1 additional non-GHG effects such as temperature inversion, aerosol/haze pollution effects, solar canyon issues, and loss of cooling and is shown to be in line with observed data in Section 3.1.10.1.

3.1.7. Dry and Humid Rural Mixed Climate Amplification Estimate: Model 5

Geometry strongly differs in rural compared to urban areas which are dominated by cities. In rural settlement areas, microclimate amplification is relatively lighter in contribution from CO₂ and water vapor feedback/GHGs for humid, sub-humid, and semiarid areas. However, there are still substantial ISAs in rural regions. Huang et al. [28] estimated that about 67% of ISAs are located in rural settlements. There are still similar amplification contributions even in light settlements besides GHGs such as loss of evapotranspiration cooling and wind reductions. We can anticipate some average amplification (i.e., >1). Therefore, we can provide a very small amplification estimate increase for rural areas, yielding.

$$\overline{A}_{Rural} = 1.1 = \overline{A}_{UHI \ mixed} / 2 \qquad \text{Rural Model 5} \tag{13}$$

3.1.8. Estimating AHR Influence on Global Warming Using Microclimate Amplification Models

While it is helpful to separate climates in terms of humid versus dry areas, now that we have a mixed Model 3 local amplification estimate value of 2.2 and footprint estimate $\overline{A}_{Footprint} = 3.2$, we can use these to assess the urbanization AHR global warming influence. Per Equations (4), (6), (11), and (12), we can now combine the local microclimate and background global amplification estimate with the AHR assessment in Equation (4). Then the global warming impact due to the AHR with background global and microclimate secondary amplification effects yields

$$%P_{AHF-GS\&LS}(2021) = \frac{P_{AHF-GS\&LS}}{5.1 \text{ Wm}^{-2}} = \frac{\{0.0356 \text{ Wm}^{-2}(A_{UHI-Mixed} \times A_{FI} \times 0.75 + 0.25 \times A_{Rural}) 2.15 \times 1.62\}}{5.1 \text{ Wm}^{-2}} = \begin{cases} 4.7\% \text{ for } A_{FI} = 1\\ 6.5\% \text{ for } A_{FI} = 1.45 \end{cases}$$
(14)

Here we insert $A_{UHI-Mixed} = 2.2$ from Equation (11), $A_{Rural} = 1.1$ from Equation (13), $A_{FI} = 1$ and 1.45 from Equation (12). The microclimate amplification factors apply mainly to UHI areas where according to the IEA [5] "with 4.2 billion people living in cities today, urban areas account for 75% of global energy consumption" (see also Yang et al. [14]) yielding the 0.75 weighting factor in Equation (14). This result finds estimates of 4.7% to 6.6% GW influence (0.24 Wm⁻²–0.34 Wm⁻²) due to AHR, when both global and microclimate UHI amplifications are considered. Results are highly dependent on the amplification estimates and are independent of the urbanization area.

The results found as much as 6.5% GW influence, yielding a 0.92% decade⁻¹ increase; this energy consumption estimated GW effect is from 1950 to 2022 and as expected correlates strongly to the population growth rate (Feinberg, [3]). This is evident from the energy percent increase in 2021 of 1.2% (=2133 TWh/176,431 TWh) which is close to the population growth rate that has decreased in the last 70 years from 2% to 0.89% (https://www.worldometers.info/world-population/, last accessed on 11 January 2023). Since the population growth rate is slowing down, likely, the energy consumption rate may eventually decrease somewhat. However, it is clear from the energy consumption GW effect, that the population growth rate is too high and is still the key issue [3].

3.1.9. Area and AHR Local Baseline Heat Flux Estimates

Results were obtained by averaging the anthropogenic warming energy over the entire earth. However, it is likely most of this AHR occurs in cities where 55% of the world's population lives [4]. IEA [5] and Yang et al. [14] estimated that 75% of energy is consumed within cities. In this and the next section, a possible scenario illustrates how modeling estimates can be shown to be self-consistent with the final results in Section 3.2.2 and agree with the main data observation by Zhao et al. [16] and other authors supporting the models.

To estimate heat flux and likely amplification effects it is helpful to overview some of the literature observations. Pigeonet et al. [31] investigated the AHR in the European agglomeration of Toulouse in France. They found that vehicle traffic was the major source of the AHR during summer. During winter, the AHR reached 100 Wm⁻² in the densest areas, whereas it ranged between 5 and 25 Wm⁻² in suburban areas. Chen et al. [32] determined, "The annual mean AHR flux in concentrated regions reached 20 W m⁻² in 2009". Bohnenstengel et al. [8] in London found values in the city center exceed 400 Wm⁻² and values at the fringes of London are of the order of 60 Wm⁻² in December. Narumi et al., 2009 studied heat release quantities in a Japanese megacity up to 48 Wm⁻² and showed that calculated AHR results were similar to observed atmospheric ΔT values.

To obtain an average AHR baseline urban heat flux estimate, it is helpful to take a lower value. Here we will use a possible scenario for a low AHR baseline heat flux of 6.5 Wm^{-2} . The reason to select this value will be clear in Section 3.1.10.1. Its absolute value is not the main goal as estimates focus on difference data in Section 3.1.10.1. However, the value simplifies difference data estimates and shows self-consistency in models and results. It is also close to the Pigeonet et al. [31] 5 Wm⁻² lower measurement. It is reasonable to

use a low and likely unamplified heat flux baseline estimate. This is then the target goal in Equation (15).

To obtain an idea of how 6.5 Wm^{-2} could occur we can consider Equation (4) results for an estimate of the increase in anthropogenic waste heat flux in UHIs. To do this, we need an estimate of the UHI area. Feinberg [25,53] using the definition of what is urban by Liu et al. [54], showed the portion of the earth that has been urbanized in this time period is 0.28%. Other authors Zhao et al. [55], for example, reported about 0.2% in 2020. Huang et al. [28] reported 0.994 million km² (0.2% of the earth) using 2016 mappings.

Sun et al. [27] estimated impervious surface areas where "we created spatially explicit and temporally consistent global maps of impervious surface area at 10-m spatial resolution for 2015 and 2018, namely the high-resolution global impervious surface area (Hi-GISA) maps . . . The total area of global impervious surfaces grew from 1.27 million km² in 2015 to 1.29 million km² in 2018" (about 0.255% of the earth).

Theobald et al. ([56] questioned the amount of earth that has been transformed by human modification from 1990 to 2017. Theobald et al. [56] made two key estimates: "between 1990 and 2015 resulted in 1.6 million km² of natural land lost, and a contemporary (\sim 2017) estimate of human modification that included additional stressors and found that globally 18.5 million km² of lands have been modified. However, a breakdown for impermeable surfaces was not easily estimated from their data but shows concern for higher estimates.

We see that there is a wide range of areas of the earth that is modified and subject to interpretation regarding urbanization. It is of interest to average the value of Equation (4) over the urbanization areas. Therefore, in this assessment, we can look at several urbanization area values (see Tables 4 and 5).

As an initial calculation, first, consider the urbanization estimate of 0.255% due to Sun et al.'s [27] for ISAs. We use an adjusted estimate from Huang et al. [28] of about 40% for ISAs in urban areas that are likely dominated by UHI building sides (see Section 3.2.1.1 discussion). This is a 0.4-factor multiplier in this area. The actual UHI built-up areas for AHR are larger when one takes into account tall buildings with small multiple rooms of apartments and office spaces each dissipating heat fluxes due to heating and cooling energy use. We can account for this in Equation (15) with an f factor. Here an adjustment factor of f = 4.0 is used as the area increase which provides the desired scenario for the baseline values $P_{AHR-UHI} = 6.5 \text{ Wm}^{-2}$. In general, exact estimates are not a major concern. This rough approximation will provide temperature difference estimates in the next section which are not very affected by the absolute value. As we referenced earlier, 75% of AHR energy is consumed in cities (IEA [5] and Yang et al. [14]). We can then incorporate this energy factor in the numerator and area factors in the denominator to obtain a possible scenario for the baseline heat flux

$$P_{AHF-UHI}(2021) = 0.0356 \text{ Wm}^{-2} \times 0.75 / (0.00255 \times f \times 0.4) = 6.5 \text{ Wm}^{-2}$$
(15)

Note that averaging over smaller areas would cause a substantial increase in anthropogenic heat flux estimates in Equation (15). Other areas are given in Tables 4 and 5. The goal here is not to obtain exact values but to provide an idea of a possible scenario illustrating how this heat flux could occur and obtain AHR ΔT difference data feasible attribution estimates in the next section which are not very effected by the accuracy in Equation (15) as will be shown. This will help support amplification modeling and their estimates as we will see in Section 3.1.10 (which will be further supported by humidity data from Zhao et al. [16] and other authors as detailed in Sections 2.2 and 3.1.4 and Appendix D).

3.1.10. Supporting Amplification Models 1–4 in Dry and Wet Environments with UHI ΔTs Estimates

We can estimate ΔT_{UHI} attribution values due to AHR in dry and wet environments using amplified baseline values and help support modeling and their values as well. Consider the possible baseline heat flux scenario in Equation (15) estimate of 6.5 Wm⁻². Although

this is not an exact estimate, ΔT attribution is based mainly on amplification estimated according to Equation (A1) and differences which are not very affected by Equation (15) absolute value. Using the resulting value, the estimated dry Model 1 amplification 1.2factor (Equation (8)) provides an assessment of the increase in the baseline heat flux for Equation (15) due to local GHGs in UHI areas as

$$1.2 \times 6.5 \,\mathrm{Wm^{-2}} = 7.8 \,\mathrm{Wm^{-2}} \tag{16}$$

Such values, for example, $\Delta \overline{P}_U = 7.8 \text{ Wm}^{-2}$ are not unreasonable as discussed in the prior section.

From Appendix A, Equation (A2), we find that $\Delta T_{AHR-UHI_Dry} = 1.3 \text{ °C}$ where

$$\Delta T_{U} = \left[\frac{\left(\Delta \overline{P}_{AHF_UHI} + \sigma(\overline{T})_{Ambient}^{4}\right)}{\varepsilon_{eff}\sigma}\right]^{1/4} - \overline{T}_{Ambient} = 1.3 \,^{\circ}\text{C}$$

$$\Delta \overline{P}_{AHF_UHI} = 7.79 \,\,\text{Wm}^{-2}, \overline{T}_{Ambient} = 24 \,^{\circ}\text{C}$$
(17)

This result is shown in the next to last column in Table 4. For the baseline estimate (Equation (15)) we find that $\Delta T_{\rm U} = 1.09$ °C using Equation (A2) where

$$\Delta T_{U} = \left[\frac{\left(\Delta \overline{P}_{AHF_UHI} + \sigma(\overline{T})^{4}_{Ambient}\right)}{\varepsilon_{eff}\sigma}\right]^{1/4} - \overline{T}_{Ambient} = 1.085 \,^{\circ}\text{C}$$

$$\Delta \overline{P}_{AHF_UHI} = 6.5 \,\text{Wm}^{-2}, \overline{T}_{Ambient} = 24 \,^{\circ}\text{C}$$
(18)

Comparing Equations (17) and (18), results in this assessment indicate that AHR creates a temperature increase in UHIs, this is the baseline result yielding a 1.09 °C rise according to Equation (18). Then GHGs raise this to 1.3 °C as found in Equation (17) yielding a difference of 0.215 °C increase. Therefore, GHGs are responsible for 17% (=0.215/1.3) of the temperature increase in Equation (17) (shown in Col. K, Table 4). Note the stability of the last column percentages even though the baseline data is varying in Col. 3 illustrating the insensitivity of difference data to the absolute values in Col. 3 (i.e., Equation (15)).

These ΔT temperature increases due to GHG and WVF attribution effects in dry and wet climates are shown in Col. J in Table 4 with different areas.

Table 4. Physics-based model estimates for dry (Model 1) and humid (Model 2) climates.

ISA %Areas *	Climate	Baseline P _{AHR-UHI} Wm ⁻² Equation (15)	Baseline $\Delta \overline{T}_{UHI}$ °C Equation (17)	Dry Amp. Equation (8)	Wet Amp. Equation (9)	$\begin{array}{c} \mathbf{A_U}\\ \text{Wet \& Dry}\\ \text{Combined}\\ \mathbf{E}\times\mathbf{F} \end{array}$	$\begin{array}{c} AHR\\ P_{AHR-LS}\\ Wm^{-2}\\ D\times G \end{array}$	Estimate AHR $\Delta \overline{T}_{UHI}$ °C Equation (16)	GHG-WVF Amp. Au ΔT Percent Effect
Α	В	D		Е	F	G	н	J	К
			Model	l (Dry)					
0.22	Dry	7.53	1.26	1.2	1	1.2	9.03	1.5	17%
0.255	Dry	6.5	1.085	1.2	1	1.2	7.79	1.3	17%
0.33	Dry	5.02	0.84	1.2	1	1.2	6.02	1.01	17%
			Model	2 (Wet)					
0.22	Humid	7.53	1.26	1.2	2.125	2.55	19.2	3.17	<u>60%</u>
0.255	Humid	6.5	1.09	1.2	2.125	2.55	16.6	2.74	60%
0.33	Humid	5.02	0.84	1.2	2.125	2.55	12.8	2.13	61%

* Estimate UHI area with building area increases (see Equation (15)).

Using the method of Equation (11), we can estimate the mixed climate estimate for the last column for *GHG-WVF* attribution finding

$$A_{GHG-WVF}(\Delta T) = (0.33 \times 17\% + 0.67 \times 60\%) 1.05 = 48\%$$
⁽¹⁹⁾

This result suggests the weighted average temperature increase attributed to *GHG* or *WVF* is about 48%.

First notice the difference in dry climates between the estimated AHR ΔT_{UHI} and the baseline ΔT_{UHI} : 1.5 – 1.26 °C = 0.25 °C, 1.3 – 1.09 °C = 0.216 °C, and 1.01 – 0.85 °C = 0.167 °C. This averages 0.21 °C due to Model 1 GHG (mainly CO₂ increase). In Appendix D, a study by Liu et al. [57] is discussed related to the COVID lockdown period that studied over 300 megacities in China where traffic was minimal and showed an average daynight reduction in surface temperature of 0.24 ± 0.22 K and the canopy of 0.41 ± 0.27 K representing about a 25% and 40% reduction, respectively. We see that surface temperature is likely more related to CO₂ GHGs while the canopy is likely more related to AHR and the GHG effect. The surface temperature difference averaging 0.21 °C provides a nice level of confidence in the Model 1 estimate which is used in Table 2 for the dry climate with the 1.2 factor related to CO₂ and yielding a somewhat lower estimate compared to Liu et al. [57] surface temperature change observation of 0.24 °C, due to a likely reduction in CO₂ emission because of lockdown.

Next note the differences in ΔT s between dry and wet climates: $3.17 - 1.5 \degree C = 1.67 \degree C$, $2.74 - 1.3 \degree C = 1.44 \degree C$, and $2.13 - 1.01 \degree C = 1.12 \degree C$. This averages $1.41 \degree C$. Zhao et al. [16] observed on average that wet climates were 3.3 K higher than dry climates. This indicates modeling is not completed yet and we can improve estimates as detailed in the next section.

3.1.10.1. Further Supporting Models 1-4 in Dry and Wet Environments

There are two main reasons that the dry-wet climate difference was only 1.2 °C in Table 4, yielding 2.1 °C lower than what Zhao et al. [16] observed. In Column G, the amplification values A_u could be higher. Another key reason is that ΔT in dry and wet climates also depends on solar heating of ISAs in UHIs. We can provide a likely scenario to make up for the missing heat fluxes from solar heating of ISAs in UHIs in Tables 5 and 6 by increasing the baseline estimate proportionately to the anticipated increase due to global warming percentages found in Table 7 bottom row. This will aid in demonstrating self-consistency in modeling alignment with data. The additional GW% from urban ISAs is 4% (bottom row, Col. 10, Table 7). Note also in the bottom row the 6.5% value (Col. 8, Table 7 where 6.5% goes as 6.5 Wm^{-2}). This prompted the simple proportionate value of 6.5 Wm^{-2} used in Equation (15). Now we are in a position to add another proportionate baseline value in Col. D Table 4 of 4 Wm^{-2} (Col. 10, Table 7 where 4% goes as 4 Wm^{-2}) to each value in Col. D Table 4, yielding the Table 5 Col. D values. That is for self-consistency with ISA and AHR global warming data, we can provide proportionate percentages to heat fluxes values. For example, the total baseline heat flux Equation (15) would increase to

$$6.5 \,\mathrm{Wm^{-2}} + 4 \,\mathrm{Wm^{-2}} = 10.5 \,\mathrm{Wm^{-2}} \tag{20}$$

shown in data Row 5, Table 5. We next increase the amplification factors by $A_{FI} = 1.45$ factor in Equation (12), Model 4 to illustrate its effect and how it is justified in this scenario. We can see that it will help to simulate the Zhao et al. [16] observations using the A_{FI} factor. This is carried out in Table 5.

ISA% Areas	Climate	Baseline P _{AHR-UHI} Wm ⁻² Equation (19)	Baseline ΔT _{UHI} °C Equation (17)	Dry Amp. Equation (8)	Wet Amp. Equations (9) and (12) xA _{FI} = 1.45	$\begin{array}{c} \mathbf{A}_{U}\\ \text{Wet \& Dry}\\ \text{Combined}\\ \mathbf{E}\times\mathbf{F} \end{array}$	$\begin{array}{c} P_{AHR\text{-}LS} \\ Wm^{-2} \\ D \times G \end{array}$	$\Delta \overline{T}_{UHI} \\ {}^{\circ}C$ Equation (16)	GHG-WVF Amp. Au ΔT Percent Effect
Α	В	D		Е	F	G	Н	J	К
			Model	1 (Dry)					
0.22	Dry	11.53	1.92	1.2	1 imes 1.45	1.74	20.1	3.31	42%
0.255	Dry	10.5	1.75	1.2	1 imes 1.45	1.74	18.3	3.02	42%
0.33	Dry	9.02	1.50	1.2	1 imes 1.45	1.74	15.7	2.61	42%
	5		Model	2 (Wet)					
0.22	Humid	11.53	1.92	1.2	2.125×1.45	3.7	42.6	6.92	72%
0.255	Humid	10.5	1.75	1.2	2.125×1.45	3.7	38.8	6.32	72%
0.33	Humid	9.02	1.50	1.2	2.125 imes 1.45	5.7	33.4	5.45	72%

Table 5. Physics-based model estimates for dry (Model 1) and humid (Model 2) climates.

Note now the differences in ΔT s between dry and wet climates are 6.92 – 3.31 °C = 3.6 °C, 6.32 – 3.02 °C = 3.29 °C, and 5.45 – 2.61 °C = 2.85 °C. The average of the differences is 3.25 °C which is very close to what Zhao et al. [16] observed that UHIs in wet climates are 3.3 K higher than UHIs in dry climates. Furthermore, note the 0.255% area value provides the best results, this somewhat supports Sun et al.'s [27] area estimate. Table 5 results are due to the added likely scenarios in heat flux from unshaded ISAs and increasing the amplification by A_{FI} = 1.45 the Model 4 factor in Equation (12). Note also that the likely scenarios of ΔT values are not uncommon for UHI observations. Indeed, a study by climate central [58] found in 159 cities in the US, that complex UHI could be much higher with 8–11 °C ΔT values. The values in Table 5 then illustrate likely possible data scenarios.

This helps justify Equation (12), Model 4 footprint assessment as well it supports Model 1 and 3 amplification values by finding reasonableness with data observation including Zhao et al. [16] and self-consistency in results in Table 7 since these models are also used in Equation (14) and Tables 6 and 7. Given the difficulty of providing physics-based amplification estimates, such data-aided verification is extremely helpful in proving modeling confidence.

3.2. Part II: Impermeable Surfaces Albedo Land Cover Change Effects

Considering the AHR GW% estimate ranging from 4.7% to 6.5% averaging 5.6% in Equation (14), the discrepancy with Zhang et al. [16] assessment is

$$12.7\% - 5.6\% = 6.9\% \tag{21}$$

Therefore, about 6.9% yields a leftover urbanization global warming likely uncertainties due to other issues, primarily, the urbanization land cover albedo effect (Feinberg, 2020), with various contributions differing from city to city. This is the topic of Part II.

3.2.1. GW Effect due to Solar Heating of ISAs with Secondary Effects

It is common knowledge and many papers have reported high land surface contact temperatures (LSCT) due to solar heating of roads and roofs. Rasmussen et al. [59] reported results of NASA ECOSTRESS found in Las Vegas pavement areas where temperatures were up to 27 °C hotter than non-ISA. Pavements are often so hot they cause health risks in contact burns (Kowal-Vern et al. [60]) as well hot pavements cause problems for dogs [61]. Interestingly, air temperatures quickly become uncorrelated with surface temperatures at an altitude of 5 feet. Morabito et al. [62] estimated about 0.4 °C to 1 °C air temperature rise for every 20% increase in impermeable surfaces near building areas. Many authors [1,63] measure the surface UHI ΔT as $\Delta T_{\text{Surface UHI}} = \text{LSCT}_{\text{built-up}}\text{-LSCT}_{\text{rural}}$. However, contact measurements on high-temperature pavements (>5 °C) between rural and built-up areas can be similar to the ambient differences.

Estimates are provided for albedo and LSCT of impermeable surface areas due to solar heating and the solar 'energy in' absorbed. Using the LSCT method is helpful as UHIs in megacities have large solar ISAs on building sides with windows and solar canyon effects that satellite albedo measurements are unable to assess and are not considered in albedo mappings (see Section 4.2). 'Energy in' is a key factor in assessing the ISA for urbanization energy flux. In this approach, we can study solar heating of unshaded ISA.

In this section, impermeable surface forcing is assessed using contact temperature difference (C Δ T) with albedo estimates (see Table 6). LSCT assessments above ambient, are a helpful practical additional ISA metric that we are familiar with as pavement temperatures can become very high.

Mentaschi et al. [62] studied surface urban heat island temperatures and found, "Within many cities, there are hotspots where extreme surface UHI intensity is 10–15 °C higher compared to relatively cooler city parts." This is mild compared to the 28 °C difference estimated in the NASA observation [59] and 32 °C by Knox [64] It is also easy to measure, the author [65] using a heat gun observed older asphalt road temperatures in suburban areas were about 51 °C and new roads at about 53 °C taken at an ambient temperature of 33 °C and concrete about 44 °C. This is 20 °C higher than the ambient for asphalt and 11 °C higher for]. One may observe that new road albedos may not age as much as old roads since surfaces appear smoother so they likely clean better in the rain. This is unfortunate as the surfaces are likely to maintain higher temperatures compared to older roads. Note the difference that concrete is 13 °C cooler than asphalt. At cooler ambient of 12 °C, results found that pavements were 7 °C to 10 °C higher than ambient depending on the time of day and if the streets were surrounded by buildings.

To help determine the impermeable surface energy dissipated from solar heating, we can use the global ISA estimated by Sun et al. [27] as discussed in Section 3.1.9 found 1.29 million km² (\sim 0.255%) in 2018.

3.2.1.1. Irradiance and ISA Percentages Adjustment Modeling Factors

To adjust for UHI solar irradiance (F-Solar UHI) and urban ISA percentages, several factors apply where

$$F - Solar UHI = f1 \times f2 \times f3 = 0.47 \times 0.65 \times \begin{cases} 0.33 \\ 0.41 \\ 0.5 \end{cases} = \begin{cases} 0.1008 \\ 0.1253 \\ 0.1528 \end{cases}$$
UHI Factor (22)

The factors are as follows: f1 = 0.47 accounts for the global average irradiance through clouds [24], and f2 = 0.65 accounts for the unshaded UHI areas which are higher than 50% to account for the solar canyon effect. The shade factor is discussed below. Lastly, the value f3 is the fraction of urban ISA likely dominated by city building sides. Huang et al. [28] report ISA-weighted averages of 33% urban and 67% rural. The urban ISA seems underweighted considering that there are 540 million buildings in the world as tagged in the OSM database (https://taginfo.openstreetmap.org/keys, last accessed on 29 December 2022). Huang et al. (2022) use a complex algorithm for building sides and provided assessment errors (also see Section 4.2). We might best use a variety of values to adjust for the possible underweighted error. Additionally, the definition of urban and rural is not easy to define. Therefore, within the Huang et al. [28] margin of error for urban ISAs, we consider values 50%, 41%, and 33% for the urban ISA percent as shown in Equation (22).

Outside the UHI area, an adjustment factor is needed for rural irradiance and ISA percentage estimates, similar to Equation (22) as follows

$$F - Solar UHI = f1 \times f2 \times f3 = 0.47 \times 0.75 \times \begin{cases} 0.67 \\ 0.59 \\ 0.5 \end{cases} = \begin{cases} 0.2362 \\ 0.208 \\ 0.1763 \end{cases} Rural (23)$$

The rural factors are as follows: f1 = 0.47 accounts for the global average irradiance through clouds ([24] and f2 = 0.75 accounts for unshaded areas where it is anticipated that less shade occurs compared to UHI tall building effects. Lastly, we use a variety of ISA rural area fractions for f3 in Equation (23) based on Equation (22) values (1-f3_{urban}).

3.2.1.2. Global Warming Estimates due to Solar Heating of ISAs

Table 6 provides global warming (GW) influences at various LSCT changes denoted as C ΔT , above ambient (of 14.5 °C) increase in Col. A. The actual LSCT is then 14.5 °C + C ΔT . Initial values in this table use the average ISA estimates of 41% urban and 59% rural (Equations (22) and (23)). Table 7 provides the alternate values in Equations (22) and (24). An example calculation for Table 6 is provided in the next section.

Note the albedo estimate (Col. B) from the LSCT in Col. A. At 0 °C above ambient, the albedo is 0.3 which is taken as the average reflectivity of the earth's system and this requires we use the average surface temperature of the earth of 14.5 °C for the albedo reference point (see Equation (25)). It is important to note several key albedo estimates of 0.15, 0.133, and 0.115 are highlighted. These are close to values reported by Sugawara and Takamura [66] of 0.12 for the cities in Japan they investigated, Tricia et al. [67] also found a

value of 0.12 for UHI urbanization Boston areas. As well Guo et al. [19] reported average albedo values in the range of 0.15 in China.

Consider the LSCT above ambient value C ΔT of 9 °C (albedo value of 0.15). The upper area uses a local amplification factor from Model 3 of 2.2. This yields a 2.4% urban global warming influence dominated by cities. We would anticipate that rural ISAs have higher average LSCT as they have about 20% more asphalt roads [28]. Therefore, we selected an increased C ΔT of 10 °C (albedo of 0.133) yielding a rural contribution of 2.3%. Then the total ISA contribution is 4.7% (see data row 4 in Table 7) over the approximate time period of 1950 to 2019. Table 7 provides an estimate for different ISA areas, urban-to-rural area ratios, and local amplification estimates.

Table 6. Effect of solar heating with estimates of ISTs and its likely global warming impact.

LSCT C∆T Average above Ambi- ent 14.5 °C	Albedo Average Estimate At LSCT * Equation (A7)	Solar Heating ISA Forcing Wm ⁻² Equation (A1)	Total Solar Heating ISD Watts (Using ISA 1.3 × 10 ⁶ km ²)	Col. D Div by Area of Earth Wm ⁻²	Col. E Div by 5.1 Wm ⁻² %GW As in Equation (5)	Global Amp Factor 1.62 × 2.15 Table 3	F- Solar UHI	F-Solar Rural	Local Amp. Factor Equations (11) and (12)	Local Rural Amp A _{Rural} Equation (13)	UHI Global % F × G × H × J	Rural %GW F × G × I × K
Α	В	С	D	Е	F	G	Н	I	J	K	L	Μ
				GHG	-WVF Amp	lification (Col. J)					
0	0.300	0.000	0.000	0.000	0.00%	3.48	0.1253	0.208	2.2	1.1	0.0%	0.0%
1	0.284	5.43	7.05×10^{12}	0.014	0.27%	3.48	0.1253	0.208	2.2	1.1	0.3%	0.2%
5	0.219	27.70	$3.60 imes10^{13}$	0.071	1.38%	3.48	0.1253	0.208	2.2	1.1	2.3%	1.1%
8	0.168	45.02	$5.85 imes 10^{13}$	0.115	2.25%	3.48	0.1253	0.208	2.2	1.1	2.2%	1.8%
9	0.150	50.91	$6.62 imes 10^{13}$	0.130	2.54%	3.48	0.1253	0.208	2.2	1.1	2.4%	2.0%
10	0.133	56.87	$7.39 imes10^{13}$	0.145	2.84%	3.48	0.1253	0.208	2.2	1.1	2.7%	2.3%
11	0.115	62.88	$8.17 imes10^{13}$	0.160	3.14%	3.48	0.1253	0.208	2.2	1.1	3.0%	2.50%
12	0.097	68.95	$8.96 imes10^{13}$	0.176	3.45%	3.48	0.1253	0.208	2.2	1.1	3.3%	2.74%
13	0.079	75.08	$9.76 imes 10^{13}$	0.191	3.75%	3.48	0.1253	0.208	2.2	1.1	3.5%	2.90%
14	0.061	81.28	$1.06 imes10^{14}$	0.207	4.06%	3.48	0.1253	0.208	2.2	1.1	3.9%	3.2%
				Foot	print Ampli	fication (O	Col. J)					
0	0.300	0.000	0.000	0.000	0.00%	3.48	0.1253	0.208	3.2	1.1	0.0%	0.0%
1	0.284	5.43	7.05×10^{12}	0.014	0.27%	3.48	0.1253	0.208	3.2	1.1	0.4%	0.2%
5	0.219	27.70	$3.60 imes10^{13}$	0.071	1.38%	3.48	0.1253	0.208	3.2	1.1	1.9%	1.1%
8	0.168	45.02	$5.85 imes 10^{13}$	0.115	2.25%	3.48	0.1253	0.208	3.2	1.1	3.1%	1.8%
9	0.150	50.91	$6.62 imes10^{13}$	0.130	2.54%	3.48	0.1253	0.208	3.2	1.1	3.5%	2.0%
10	0.133	56.87	$7.39 imes10^{13}$	0.145	2.84%	3.48	0.1253	0.208	3.2	1.1	4.0%	2.3%
11	0.115	62.88	$8.17 imes10^{13}$	0.160	3.14%	3.48	0.1253	0.208	3.2	1.1	4.4%	2.50%
12	0.097	68.95	$8.96 imes10^{13}$	0.176	3.45%	3.48	0.1253	0.208	3.2	1.1	4.8%	2.74%
13	0.079	75.08	$9.76 imes 10^{13}$	0.191	3.75%	3.48	0.1253	0.208	3.2	1.1	5.1%	2.90%
14	0.061	81.28	$1.06 imes10^{14}$	0.207	4.06%	3.48	0.1253	0.208	3.2	1.1	5.7%	3.2%

* LSCT = $14.5 \circ C + C\Delta T$.

3.2.1.3. Example Calculation

As an example of the calculations in Table 6, Consider the highlighted footprint data row in Col. A where $C\Delta T = 10$ °C (data row 16), yielding an LSCT temperature of 14.5 °C + 10 °C = 24.5 °C. From Appendix A, Equation (A1), we find the results in Col. C of

$$\Delta \overline{P}_{Forcing} = \varepsilon_{eff} \sigma \left(\left(\overline{T}_{Ambient} + \Delta T_{U} \right)^{4} - \left(\overline{T} \right)^{4}_{Ambient} \right) = \sigma \left(\left(14.5 + 10 + 273.15 \right)^{4} - \left(14.5 + 273.15 \right)^{4} \right) = 56.86 \, \mathrm{Wm^{-2}}$$
(24)

In Column A, a C ΔT of 10 °C yields the albedo value using Appendix B, Equation (A7) estimated as

$$(\Delta \alpha) = \varepsilon_{eff} \sigma (\overline{T}_{S}^{4} - (\overline{T}_{S} + \Delta T)^{4}) / \frac{S_{o}}{4} = 56.9/340.25 = 0.167$$
(25)

where $\alpha_S = 0.3 - \Delta \alpha = 0.3 - 0.167 = 0.133$. Note Equation (25) suggests a 10 °C variation for an albedo change of 0.167 or about 0.0167/°C. Heat gun observations from the author (Section 3.2.1) found a 13 °C change between the old concrete and old asphalt. The differences in albedo vary depending on aging and materials, roughly between old asphalt

and concrete is $\Delta \alpha = 0.18-0.21$. This would yield about $0.014/^{\circ}$ C to $0.0162/^{\circ}$ C. This is fairly close to theory. Note that $\varepsilon_{eff} = 1$ was used and a better value is closer to 0.95 which would yield a small correction in the tables.

We can use the urbanization (urban and rural) area due to Sun et al. [27] in Col. D, $56.9 \text{ Wm}^{-2} \times 1.3 \times 10^{12} \text{m}^{-2} = 7.39 \times 10^{13}$ Watts, dividing this by the area of the earth yields a Col. E value of 0.145 Wm^{-2} ; then dividing this by 5.1 Wm^{-2} (see Equation (7)) yields 2.84% in Col F. We then multiply this by the background global amplification factor of 2.15×1.62 , the microclimate amplification factor of 3.2 (Equation (12)), the solar UHI adjustments irradiance factor of 0.1253 (Equation (22)) in Col. H yielding Col. L results of (data row 20)

Urban estimate =
$$2.84\% \times 2.15 \times 1.62 \times 3.2 \times 0.1253 = 4.0\%$$
 (26)

However, the rural average unshaded LSCT is estimated to be higher with more open asphalt roads of 11 °C, this row yields 3.14% before applying factors which then finds (data row 21, Table 6)

Rural Estimate =
$$3.14\% \times 2.15 \times 1.62 \times 1.1 \times 0.208 = 2.5\%$$
 (27)

Then the total estimate has an average GW influence of about 6.5%. This value is given in the eighth data row in Table 7. Huang et al. [28] reported that roads consist of about 14% of the ISA. This would indicate that about 6.5% \times 0.14 = 0.91% of global warming is possibly due to roads (shown in Table 7). However, the likely value of 1.1% (see next section) is discussed in the next section. Other estimates are provided in the next section.

In this calculation, the adjustment factors for shade are not well known but estimates provided help in understanding feasibility. This assessment is less straightforward than the estimate for AHR influence in Equation (14) because of the adjustment factors in Equations (22) and (23). However, these factors are believed to be reasonable. Such modeling is likely more conservative than climate models related to GHGs such as CO₂ estimates on GW that do not include UHF contributions as detailed in this study (see Appendix F).

Note the 6.5% GW influence is the best estimate and helps assess the root causes of Zhang et al. [1] result (Table 7).

It is interesting to note that Guo et al. [19] in China found in 10 of the 11 cities they assessed had albedo values that averaged around 0.15–0.16. They concluded the cities had an overall cooling effect. By comparison, the 0.15 albedo value in Table 6 yields on a global scale urban value of about a 3.5% GW influence. Note in this study comparisons are made by assessing ISA temperatures that are above ambient. This allows us to work relative to earth ambient estimates. Results show the need to add city amplification effects when deducing land cover albedo influences.

3.2.2. Combined Results

From Table 7, the likely scenario for high-temperature unshaded impermeable surfaces (including building sides) averages around 10 °C for urban and 11 °C for rural, above a 14.5 °C ambient (with albedo ISA estimates of 0.133 and 0.115 respectively). This likely scenario considers some of the reported city albedos in the range of the estimates of 0.115 to 0.16 yielding a total UHF contribution of 13% that matches the Zhang et al. [1] assessment. This yields a measure of modeling self-consistency along with the data-aided modeling results in Section 3.1.10. Using these averages, the ISAs are estimated to contribute 6.5% to global warming when considering the footprint 3.2 and rural 1.1 amplification estimates (see last row Table 7). This is broken down with urban ISA contributing an average of 4.0% and rural areas contributing about 2.5% GW influence.

UHI LSCT CΔT (°C) above Ambient (%ISA)	UHI ISA Albedo Value	Rural LSCT CAT (°C) above Ambient (%ISA)	ISA Albedo Rural Values	Local Amp UHI	Rural Local Amp	GW% from AHR	GW% from All ISA	GW% from ISA of Roads	Urban ISA GW% (Rural ISA GW%)	Total ISA & AHR GW%
			G	HG-WVF	Amplifica	ation				
10 (50%)	0.133	11 (50%)	0.115	2.2	1.1	4.74	5.4	0.76	3.3 (2.1)	10.1
10 (41%)	0.133	11 (59%)	0.115	2.2	1.1	4.74	5.2	0.73	2.7 (2.5)	9.95
10 (33%)	0.133	11 (67%)	0.115	2.2	1.1	4.74	5.0	0.7	2.2 (2.84)	9.8
			F	ootprint	Amplificat	tion				
9 (50%)	0.15	10 (50%)	0.133	3.2	1.1	6.5	6.2	0.87	4.3 (1.9)	12.7
9 (41%)	0.15	10 (59%)	0.133	3.2	1.1	6.5	5.8	0.81	3.5 (2.3)	12.3
9 (33%)	0.15	10 (67%)	0.133	3.2	1.1	6.5	5.5	0.77	2.9 (2.6)	12.0
10 (50%)	0.133	11 (50%)	0.115	3.2	1.1	6.5	7.1	1.0	5.0 (2.1)	13.6
10 (41%)	0.133	11 (59%)	0.115	3.2	1.1	6.5	6.5	0.91	4.0 (2.5)	13.0
10 (33%)	0.133	11 (67%)	0.115	3.2	1.1	6.5	6.14	0.86	3.3 (2.84)	12.6
11 (50%)	0.115	12 (50%)	0.097	3.2	1.1	6.5	7.6	1.1	5.3 (2.3)	14.1
11 (41%)	0.115	12 (59%)	0.097	3.2	1.1	6.5	7.1	1.0	4.4 (2.7)	13.6
11 (33%)	0.115	12 (67%)	0.097	3.2	1.1	6.5	6.6	0.92	3.5 (3.1)	13.1
				Road	Estimate					
12 (41%)	0.097	12 (59%)	0.097	3.2	1.1	NA	7.6	1.1 *	4.8 (2.7)	NA
			Fo	otprint A	verage Re	sults				
10 (41%)	0.133	11 (59%)	0.115	3.2	1.1	6.5	6.5	1.1 *	4.0 (2.5)	13.0

Table 7. Combined results from Table 6 with various values.

* Evaluated at an albedo of 0.1 (see Huang et al., 2022).

Asphalt roads ISAs are estimated to have about a 1.1% global warming influence evaluated at an albedo of about 0.1 (next to the last row, Table 7). Note however, the ISA for roads is based on Huang et al.'s [28] estimate, which they detail is likely biased owing to incomplete data and lack of detection on narrow roads due to limitations of the spatial resolution. Therefore, the value in Table 7 that yields a GW influence of 1.1% may be higher.

As it was pointed out these results are not as simple as the AHR assessment in Equation (14) due to the estimates in Equations (22) and (23). Here we are reminded that climate modeling is not a perfect science and the goal here is to provide likely scenarios to illustrate support for the Zhang et al. [1] findings.

Primarily, these results help explain root causes and provide a level of confirmation for the Zhang et al. [1] results found from ground-based station temperature measurements. That is when the ISA and AHR heat flux contributions are added together the GW contribution averages about 13%. Zhang et al [1] state "the magnitude of the urbanization effect on global land averaged annual mean surface air temperature change over 1951–2018 corresponds to urbanization contribution of 12.7%". The Zhang et al. [1] study is the most comprehensive recent global measurement study and used a new method based on machine learning to classify uncontaminated observatory stations into rural and urban retaining 12,505 stations out of 33,878 worldwide.

Note that Zhang et al. (2021) observations may likely have been dominated by ISA heating compared to AHR due to the altitude of their observations at about 5–6.5 feet. This suggests possible higher warming issues due to urban areas dominated by cities. As well

Zhang et al. [1] observations were limited to UHI areas. Rural assessments in this study are significant, averaging 38% of the GW ISA influence.

3.3. Mitigating Urbanization Heat Fluxes

3.3.1. Natural Reflectivity of Land as UHI Comparison

Satellite measurements assess the impact of urbanization often making an albedo comparison to nearby rural areas which have either modest or no urbanization activities. As we point out, this is a challenging comparison since cities amplify heat making it difficult to equate it to the albedos of rural areas. One might also ask is this the best comparison when judging GW effects? In this study, the reference comparison was to consider ISA temperatures above the earth's average ambient temperature. For Satellite albedo assessments, a likely equivalent alternate comparison would be to consider the global natural average reflectivity of land or earth. He et al. [20] found that the natural land albedo varies from 0.1 to 0.4 with an average of 0.25. In Guo et al. [19] study that found substantial negative albedo-induced radiative forcing in 11 cities in China from 1980 to 2015 averaging -7.76 Wm⁻², only one of the 11 cities studied had an average albedo close to 0.25. The other ten cities averaged about 0.15. Therefore, if comparisons were made relative to the natural land reflectivity, even without amplification taken into account, a large positive albedo radiative forcing would result.

Feinberg ([50] defined albedo heat pollution (i.e., excess LSCT heat flux) in terms of the planet's natural reflectivity of land as any manmade surface or combination of surfaces with resulting lower reflectivity than 25% and with local amplification \geq 1. That is, one goal might be to match the natural reflectivity of land to reduce 'heat pollution' (i.e., improper land-cover and land-use) and UHI contributions to global warming. For example, Feinberg [50] showed that if we compare a black asphalt road to the natural average reflectivity of land (25%) with global secondary effect (no local amplification factors), one acre of black asphalt roads equates to 74,200 gallons of gasoline energy per year per acre or about 7.5 times more energy in heat pollution that a solar power plant produces per acre.

It is easy to assume that the neighboring rural albedo areas should be used in GW comparisons. However, this provides a lack of information as there is a serious energy imbalance in urbanization as indicated by UHI ΔT observations. This is partly the result of local amplified heat flux. The need for local and regional mitigation is apparent from Tables 4–7, Equation (14), and Zhang et al.'s [1] work. Reducing heat fluxes also reduces the potential for local and global amplification.

3.3.2. Estimating the UHI Solar Geoengineering Requirements

We see from Figure 2, the slope is 2196 TWh per year with similar recent increases in the last 5 to 10 years. Therefore, it is difficult to mitigate GW by reducing energy consumption. Improvements in energy efficiency and population development supported by multi-actor governance would be helpful in this area.

In terms of solar geoengineering, the likely candidate is a need to increase the reflectivity of cities. Yet due to dark rooftops and pavements, this is becoming increasingly impossible [63]. A recent MIT [68] pavement study concluded that in all U.S. urban areas, an increased temperature of 1.3 °C occurs in summer months and heatwaves are 41% more intense with 50% more heatwave days due to asphalt pavements. Yet policymakers have not added solar geoengineering albedo guidance in the Paris Agreement. However, based on the work by Zhang et al. [1] and the findings in this study, we see that such goals are urgently needed.

Studies have suggested [69] pavement UHI problems even back in 2005 [70] or earlier and there is a history of UHI authors' findings warning of likely UHI contribution to global warming (see Feinberg, [25] for a review) with no mitigating effort issued in the Paris Agreement. We can estimate feasible UHI-required goals. Although each city can differ in requirements, we can look to assess the general capability of UHIs. Solar geoengineering of UHIs has been studied and increasing the UHI albedo by about 0.1 is feasible. In a paper by Akbari et al. [69] it was estimated that "white rooftops and light-colored pavements, can increase the albedo of urban areas by about 0.1 and could potentially offset some of the anticipated temperature increase caused by global warming".

Therefore, perhaps the most practical UHI mitigating goal might be to try and accomplish an albedo increase of 0.1 as indicated by Akbari et al. [69]. UHI albedos are typically in the range of 0.1 to 0.15 [19,65]. Therefore, a 0.1 increase is also close to the 0.25 natural reflectivity of the land. We can assess cooling capabilities relative to this goal for ΔT that should result. In Appendix B, Equation (A9), we find the applicable relationship for this assessment is

$$X_{c}\frac{S_{o}}{4}(\Delta\alpha) = \frac{\varepsilon_{eff}\sigma}{\overline{A}_{UHI_mixed}}(\overline{T}_{S}^{4} - (\overline{T}_{S} + \Delta T)^{4})$$
(28)

Here *T* is the average rural neighboring climate temperature, ΔT is the average observed LSCT UHI change for an increased albedo for the $\Delta \alpha = 0.1$ goal, X_c is the average irradiance, and \overline{A}_{UHI_mixed} is from Equation (11). For example, if the average neighboring rural climate temperature is 14.5 °C, and considering an average global irradiance of 0.47 [24,53] and $\overline{A}_{UHI_mixed} = 3.2$, we can solve Equation (28) finding an LSCT UHI cooling capability for unshaded areas of

$$\Delta T_{LSCT} = -9 \,^{\circ}\text{C} \quad for \quad \overline{T}_S = 14.5 \,^{\circ}\text{C}, \quad \Delta \alpha = 0.1 \tag{29}$$

Although this is an LSCT change it relates to the UHI ΔT . Increasing an albedo of 0.1 globally in cities would likely mitigate a large portion of the UHI ΔT effect and its local and global amplification effects worldwide. Note that without considering the amplification reduction \overline{A}_{UHI} mixed, $\Delta T_{LSCT} = -2.9$ °C.

From the likely best estimate, road contributions are 1.1% of global warming but may be higher due to a lack of data [28]. If asphalt was replaced with concrete, (with a 5 times more reflective type surface), the global warming cooling would be about a factor of 5 [70] yielding a reduction of at least -5.5% in global warming (5 × 1.1%). Solar geoengineering is challenging, but cool roads and roofs are feasible. An EPA ([71] and a recent MIT [67] study found that cool roads are highly cost-effective when implemented during maintenance and construction periods.

4. Discussion

4.1. Mciroclimate Amplification Control for City Cooling Requirments

Controlling the microclimate to cool off cities is important. Although the temperature is a good gauge of heat in cities, measuring the strength of the UHF microclimate amplification is likely best measured using the city's footprint. We noted the footprint measures the combined effects of the UHF microclimate amplification effects as illustrated in Figure 1. Zhou et al. [33] found an average footprint of 3.2 times the urban area and warned that "ignoring the footprint may underestimate the UHI intensity in most cases". Yang et al. [39] in a temporal study in China showed that footprints rapidly grow at an alarming rate of about 4.4% per year, indicating the rapid spreading of 'heat pollution' and lack of microclimate control in China's cities. We presented the strength of the UHF microclimate amplification in humid environments with the help of Zhao et al. [16] study that found an average ΔT increase of 3.3 K observed in daytime hours in humid climates. This data was key in supporting Models 2, 3, and 4. We found that water vapor feedback in cities was about a factor of 2.1 times higher than the standard atmosphere (Equation (9), Model 2). We noted the importance of a study by Liu et al. [23] which found during the COVID lockdown in China cities when traffic was minimal and energy use was low, that surface UHI intensity was reduced by 25%. about 40% reduced in the canopy. This study supports the CO_2 Model 1 effect and the strength of AHR in the canopy. These strong local effects add up in cities worldwide and this study found likely significant global warming contributions in Section 3.2.2 of about 13% in agreement with Zhang et al.'s [1] measurements. These key

observations show that city planners should not ignore UHF microclimate amplification effects and respect the need for cool roads and roofs as well as the need for energy efficiency to control AHR and reduce UHF in city microclimates. We find that UHF microclimate control should be an important added part of the Paris Agreement.

4.2. Satellite Assessments

Recent satellite-based studies are finding that many cities have enough of a reflective quality that one could conclude the UHI effect may be dominated by AHR compared to the albedo effect and other issues [19,21] using satellite methods found negligible urbanization albedo forcing indicating, "Relative to 2001, the increased urban lands in 2018 produced a radiative forcing of ~0.00017 Wm⁻²". However, Feinberg [25] found that increased albedo changes from urban land from 1950 to 2019 did produce higher radiative forcing using the MCAE of the order found in Equation (21). As well, as assessments in Section 3.2 and Zhang et al.'s [1] study conflict with Ouyang et al.'s [21] results.

Yang et al. [14] found that the annual average anthropogenic heat flux in mainland China provinces averaged 0.4 W/m² and could be as high as 30–40 W/m² on a city scale. Guo et al. [19] found substantial negative albedo-induced radiative forcing from 11 cities in China from 1980 to 2015 averaging -7.76 Wm⁻². That is, they observed an albedo-induced UHI cooling effect. We note this means a substantial AHR effect greater than 7.76 Wm⁻² would be needed to create the observed UHI ΔT effect in these cities indicating that AHR must be the root cause that dominates city warming trends in the Guo et al. [19] study. For example, one of the cities reported by Guo et al. [19], Shanghai, is close to the average with a negative albedo RF of -7.26 Wm⁻² and has a UHI ΔT rise of at least 1.5–2 °C with an albedo average of about 0.16. This would require AHR forcing roughly of about 15–20 Wm⁻² (see Table 4 for example). There are many potential satellite issues that may cause difficulties in such measurements discussed in Appendix G.

5. Conclusions

We note there are reasonable uncertainties in assigning attributions in global warming estimates as climate modeling is not a perfect science. However, in this study physicsbased modeling agreed with data-aided observation for UHF scenarios, as demonstrated in Section 3.1.9, Section 3.1.10, and Section 3.1.10.1. Overall, the results found:

- AHR from 1950–2021 due to energy consumption is estimated to have a maximum GW influence of 6.5% (Equation (14)). This is mainly related to population growth. Here, we see that the energy consumption increase in 2021 was 1.2% and this is highly correlated to a population growth rate [3] that decreased to 0.89% (Section 3.1.8).
- Unshaded ISAs from 1950 to 2019 are estimated to have an average GW influence of 6.5% (Table 7). This is broken down with average contributions of 4.0% from urban ISAs and 2.5% from rural ISAs.
- Heat fluxes from unshaded ISAs and AHR combined are estimated to have an average GW influence of 13% (Table 7) over the approximate time period of about 70 years from 1950.
- The main microclimate amplification factor justified with data was 3.2 (Table 7) in Model 4 (Section 3.1.10). It is assumed that UHIs dominate urban effects.
- Unshaded ISAs that helped match ground-based observations indicated that urban ISA temperatures would likely average 10 °C above a global ambient temperature with an average albedo of 0.133, while rural ISAs were estimated at 11 °C above ambient with an average albedo of 0.115. (Rural ISAs are anticipated to have a higher temperature due to increases in the percentage of asphalt roads and roofs, Table 7).
- The ISA average breakdown was 59% rural and 41% urban (Table 7).
- GHGs with water vapor feedback were found in modeling to be a major amplifier of AHR microclimate heat fluxes increasing UHI ΔTs by about 48% (Equation (19)).
- Roads are estimated to contribute 1.1% to GW but may be higher due to a lack of data and satellite resolution (see Section 3.2.2). New roads were observed to be darker and

smoother and will likely clean better in the rain, therefore, unfortunately, will likely be much hotter over their lifetime maintaining low albedos compared to old roads. The overuse of black asphalt on roads and roofs are highly dangerous to our environment, contributing significantly to urban heat wave intensity, city temperatures, and global warming, suggesting that such practices should be banned.

- Changing roads from asphalt to concrete or similar type surface reflectance can increase their reflectivity by about a factor of 5 and reduce global warming by at least 5.5%.
- Without considering any secondary amplification effects, results indicated that AHR and solar heating of ISAs heat fluxes would equate to about 0.7% and 1% GW influence, respectively.
- A heat flux likely scenario found AHR and unshaded ISAs in cities may average 6.5 Wm⁻² and 4.0 Wm⁻² (Equation (20)) respectively totaling a 10.5 Wm⁻² baseline value and this was estimated to increase UHI Δ*T* to about 1.75 °C which could be further amplified in dry and wet microclimates to about 3 °C to 6.3 °C (Table 5), respectively.
- Given average climate conditions, it is possible to mitigate much of the UHI effect with an albedo increase of 0.1 which is anticipated to lower the average impermeable surface temperatures by about 9 °C (Equation (29)) that studies show can be accomplished with cost-effective cool roads and roofs.
- Not accounting for UHFs and their microclimate and global amplification effects may
 result in climate model attribution errors of 2XUHF influence (1 × UHF due to not
 including the urbanization influence, and 1 × UHF in overestimating the current GHG
 influence, as illustrated in Section 3.3). The suggested correction in Appendix F is
 an urbanization forcing of 0.31 Wm⁻² and with feedback influence (×2.15) yields a
 value of 0.66 Wm⁻². This results in a possible 13% urbanization warming effect that
 occurred between 1950–2019.
- The forcing estimate for UHF 0.31 Wm⁻² (Section 3.3) led to an extra increase in atmospheric water vapor averaging 204 ppm (Appendix F, Equation (13))

Results help explain feasible root causes and support Zhang et al.'s (2021) findings. Assessments were made with conservative microclimate Models 1–5 that were aided by data to help justify local climate amplification factors. There is relatively high confidence in background global amplification factors. In modeling, many conservative microclimate values were used primarily in Equations (8) (Model 3), (10)–(12), (21), and (22), so warming estimates may be higher.

We discussed the importance of mitigating the UHF effects in cities worldwide primarily by increasing city albedo to at least the natural reflectivity of land which is about 25% as reported by He et al. [20]. Other authors have suggested that a 0.1 albedo increase is feasible in UHI [69] and can be cost-effective for cool roofs and roads [67].

As well from Figure 2, it is important to reduce energy consumption which appears to have a market increase slope after 1950 showing a yearly increase in consumption of 2196 TWh per year from 1975 to 2021, and this has not slowed significantly in the last 5 to 10 years as energy use is highly correlated with population growth rates [3]. Therefore, it is difficult to mitigate this by reducing energy consumption. Improvements in energy efficiency and population development supported by multi-actor governance would be helpful in this area.

At the current rate of energy consumption, one current estimate indicates we may run out of fossil fuel by 2060 (Octopus Energy [72]).

Zhang et al. [1] and Feinberg [17,25,53] recent findings and results here are unable to support the IPCC in AR6 (2021) statement that urbanization has a "negligible impact on global annual mean surface-air warming (very high confidence)." Furthermore, this is also contrary to a long history of UHI findings by other authors (see Feinberg [25] for an overview). Strong support by the IPCC in recognizing the urbanization heat flux GW risks occurring is lacking and is badly needed as it is likely an important influence on policymakers. Policymakers for many years have not implemented any worldwide Paris Agreement albedo goals. Without IPCC support to help recognize this issue the problem of poor solar urbanization color choices and yearly increases in black asphalt roads, dark roofs, and building sides will continue which has already made the task of solar geoengineering of urbanization nearly impossible [53]. Currently, negative solar geoengineering dominates urbanization worldwide. Results also point to the conclusion that the overuse of black asphalt on roads and roofs are highly dangerous to our environment contributing to city heatwaves and higher temperatures (Sections 3.3.1 and 3.3.2), and global warming findings suggest that such practices should be banned. The author noted that new roads are smoother and clean better in the rain and will likely maintain higher temperatures over their lifetime.

The urbanization heat flux concern should be at least at the level of CO_2 mitigation, especially with city growth rates. It is equally important as a helpful solution for many reasons including the albedo mitigating advantage [9,53], its ability to quickly reduce local and partial global temperatures, and the difficulty of mitigating CO_2 . As well, energy consumption increases will make GW mitigation difficult. We suggest that the Paris Agreement albedo goals detailed in Section 4.4 and discussed in studies such as Feinberg [53], are urgently needed and should be provided as soon as possible.

Funding: This research received no external funding.

Data Availability Statement: All data is provided within this paper. There are no external sources other than the references listed.

Acknowledgments: The author would like to thank the journal Land's open access stipend for this work.

Conflicts of Interest: The author declares no conflict of interest.

Abbreviations

anthropogenic heat release (AHR), footprint (FP), global warming (GW), greenhouse gases (GHGs), heatwave (HW), impermeable surface areas (ISAs), land surface contact temperatures (LSCTs), loss of cooling (LC), methods of climate amplification estimates (MCAE), solar geoengineering (SG), urbanization heat fluxes (UHF), urban heat island (UHI), upwelling (UW), water vapor feedback (WVF).

Appendix A.

The applicable equation used for Equations (17), (18), and (24), and in Tables 4–7 results are

$$\Delta \overline{P}_{Forcing} = (\overline{P}_U) - (\overline{P}_{Ru}) = \varepsilon_{eff} \sigma \left(\left(\overline{T}_{Ambient} + \Delta T_U \right)^4 - \left(\overline{T} \right)^4_{Ambient} \right)$$
(A1)

Here the subscripts U refers to urban and Ru to rural. Solving for ΔT_U we obtain

$$\Delta T_{U} = \left[\frac{\left(\Delta \overline{P}_{AHF_UHI} + \sigma(\overline{T})_{Ambient}^{4}\right)}{\varepsilon_{eff}\sigma}\right]^{1/4} - \overline{T}_{Ambient}$$
(A2)

. ...

where $\Delta \overline{P}_{AHF_UHI} = \Delta \overline{P}_{Forcing}$

Appendix B. Albedo UHI Mitigation Assessment

For a given albedo α_1 associated with a temperature T_1 , their relationship can be expressed by

$$\frac{S_o}{4}(1-\alpha_1) = \varepsilon_{eff}\sigma T_{1S}^4 \tag{A3}$$

Given an albedo change to α_2 associated with a temperature change to T_2 we can write

$$\frac{S_o}{4}(1-\alpha_2) = \varepsilon_{eff}\sigma T_{2S}^4 \tag{A4}$$

The difference from Equations (A3) and (A4) yields

$$\frac{S_o}{4}(\alpha_2 - \alpha_1) = \sigma(T_{1S}^4 - T_{2S}^4) = \varepsilon_{eff}\sigma(T_{1S}^4 - (T_{1S} + \Delta T)^4)$$
(A5)

where

$$T_{2S} = T_{1S} + \Delta T \tag{A6}$$

Equation (A5) provides a relationship between temperature and albedo changes relative to the natural reflectivity of the Earth without clouds.

$$\frac{S_o}{4}(\Delta \alpha) = \varepsilon_{eff} \sigma (\overline{T}_S^4 - (\overline{T}_S + \Delta T)^4)$$
(A7)

where

$$\Delta \alpha = \alpha_2 - \alpha_1 \tag{A8}$$

Because the LHS = RHS in Equation (A7), we are free to study either side in terms of global warming changes. Consider now an ISA denoted as A_{ISA} , having an irradiance factor X_c and with local amplification factor A, we write the relation for the UHI mitigating albedo to its ΔT observation as

$$\Delta \overline{p}_{Forcing}(\text{watts}) = X_C A A_{ISA} \sigma \varepsilon_{eff} (\overline{T}_S^4 - (\overline{T}_S + \Delta T)^4)$$
(A9)

These results are used in Equations (24)–(27). We can divide B7 by the area of the earth to obtain the global warming influence. Equation (A9) was originally developed by the author and is an alternate form of Equation (12) in the reference by Feinberg [53].

Appendix C. Local vs. Global Water Vapor Feedback Amplification Estimate Comparisons

Appendix C.1. Global Water Vapor Feedback Amplification

Gordon et al. [30] estimated a water vapor feedback of 2.4 $Wm^{-2} K^{-1}$ at a global temperature of nearly 14 °C. For a 1 K global temperature rise, this would yield an amplification of

$$A_{Water_Vapor} = \frac{5.1 \,\mathrm{Wm^{-2}}}{5.1 \,\mathrm{Wm^{-2}} - 2.4 \,\mathrm{Wm^{-2}K^{-1}} \times 1.0 \,\mathrm{K}} = 1.81 \tag{A10}$$

Liu et al. [23] estimated a water vapor feedback of $1.6 \text{ Wm}^{-2} \text{ K}^{-1}$ at a global temperature near 14 °C. For a 1 K global temperature rise, using Liu et al.'s [23] estimate in Equation (A10), this would yield an amplification of 1.5. In this paper, the full feedback is assessed at 2.15 [9,17] which includes WVF and other issues.

Appendix C.2. Local Water Vapor Feedback Amplification

Feinberg [17] estimated a water vapor feedback of $3.4 \text{ Wm}^{-2} \text{ K}^{-1}$ at a UHI temperature of 15 °C. For a 1 K UHI temperature rise, with a comparable local heat flux used in Equation (A10), results are

$$A_{Water_Vapor} = \frac{5.1 \,\mathrm{Wm^{-2}}}{5.1 \,\mathrm{Wm^{-2} - 3.4 \,Wm^{-2}K^{-1} \,\times \, 1.0 \,K}} = 3 \tag{A11}$$

This illustrates that the local water vapor feedback is about a factor of 2 higher than the global average. Most UHIs have a ΔT increase higher than a 1 K rise. This would increase the local amplification effect.

Appendix D. UHI CO₂ Surface & Dome Re-Radiation

Koerner and Klopatek [7] estimated about 80% of CO_2 originates in urban environments which could be a strong local re-radiation amplification effect. Re-radiation from CO_2 -IR interactions at the surface and in the canopy microclimate is not well understood. However, the likelihood will be primarily correlated to IR 667 cm⁻¹ emissions and CO_2 concentration levels. CO_2 concentrations in the UHI canopy are generally much higher than in rural areas. As well, many authors studying CO_2 concentration find a high correlation to vehicle traffic emissions. The idea that CO_2 re-radiation of surface IR and AHR in the canopy urban microclimate can create amplification simply extends greenhouse gas theory.

Studies show there is a high CO_2 concentration near the surface and in the canopy areas compared to rural areas. In France, Widory and Javoy [73] measured CO_2 concentrations in Paris, its suburbs, and the surrounding open countryside, as well as from vehicles, heating sources, power stations, etc. outside Paris that averaged 415 ppm, while values in the city sometimes reached as high as 950 ppm (229% increase). These higher values were driven primarily by vehicle exhaust. In Italy, Gratani and Varone [74] measured atmospheric CO_2 concentrations at various places throughout Rome in 1995, 1998, and 2001–2004. The data they obtained pointed to the existence of a higher amount of variable CO_2 in the UHI dome area. CO_2 concentrations at the periphery of the city averaged 405 ppm and 377 ppm on weekdays and weekends, respectively, whereas at the city center, they averaged 505 ppm and 414 ppm on weekdays and weekends, respectively (about a 25% increase). In addition, there was a strong correlation between traffic density and CO_2 concentration.

George et al. [75] in Baltimore measured differences in the city's center (urban site), a suburban site, and a rural site over 5 years. Atmospheric CO_2 was significantly increased on average by 66 ppm (about 19% higher) from the rural to the urban site and air temperature was also significantly higher at the urban site (14.8 °C) compared to the suburban (13.6 °C) and rural (12.7 °C) sites. Results were related to population and associated high traffic volume.

In Mexico, Velasco et al. [76] studied a densely populated area and CO_2 's highest concentration average of 421 ppm and the lowest concentrations average of 375 ppm (12%). Velasco et al. [73] found that CO_2 concentrations were "directly related to vehicular traffic, vehicles accounted for approximately 60% of the urban CO_2 emissions.

Liu et al. (2022) found "a substantial decline in both surface and canopy UHIs over 300-plus megacities in China during lockdown periods compared with reference periods". During lockdown measures, "the surface UHII was reduced by 0.25 ± 0.22 K in the daytime and 0.23 ± 0.20 K at night. The reductions in canopy UHII during lockdown periods are more significant, reaching 0.42 ± 0.26 K in the daytime and 0.39 ± 0.29 at night. The linkage between lockdown measures and UHI variations provides new insight into the influence of this pandemic on urban climate." They further state, "It is therefore convincing that the observed UHII decreases during lockdown periods are mainly associated with reduced AHR induced by the closure of road transportation, shutdown of nonessential business and industrial activities, and restrictions of outdoor activities".

Feinberg [50] in a molecular-level study on UHI microclimates indicated the capability of CO_2 to amplify heat flux in UHIs can be significant in the urban climate and can impact warming at the global level. UHI dome-surface microclimate areas where CO_2 active pollution occurs at much higher levels combined with intense IR radiation were assessed. Estimates only required about 780 times more frequent CO_2 -IR interactions per molecule per year to achieve IPCC forcing in a 'what if' scenario. Estimates were made by considering a 16% increase in CO_2 compared to the standard atmosphere. The UHI volume assessed had a fractional area of 0.188 of the earth with a height of 100 m. Assessment appeared to indicate that only the CO_2 concentration may not be the limiting factor. It was more likely that available heat flux from anthropogenic heat release and albedo absorption in UHIs limits the amount of local CO_2 re-radiation that can intensify microclimate warming in the UHI and its related effect on global warming.

29 of 34

Appendix E. Model 3 Humidity Correction for Dry Areas

The humidity correction for the dry region estimate of 33%, actually consists of 11% sub-humid and 15% semiarid; these conditions are detailed in the equation below.

 $\overline{A}_{UHI_mixed} = 0.11A_{U-Subhumid} + 0.15A_{U-SemiArid} + 0.07A_{U-Arid} + 0.67A_{U-Wet}$ $A_{U-Subhumid} = 2.125 \times 0.83 = 2.54, A_{U-SemiArid} = 2.125 \times 0.5 = 1.28, A_{U-Arid} = 1$ $\overline{A}_{UHI_mixed} = (0.11(1.76) + 0.15(1.0625) + 0.06(1) + 0.67(2.125))1.2$ = (0.1936 + 0.1594 + 0.07 + 1.424)1.2 = 2.2(A12)

The weighting factors for subhumid are due to NOAA climate ratios [77]. The chart provided in this reference finds an average P/E_p ratio of 1.2 for humid, 1 for subhumid, 0.6 for semiarid and we use arid for dry areas. Here P is precipitation (in mm) and E_p is the potential evapotranspiration (in mm). We then normalize this for humid areas yielding 1 for humid, 0.83 for subhumid, and 0.5 for semiarid. These are shown in Equation (A12). Note the 2.125 factor is from Model 2 and the 1.2 factor is from Model 1 Equation (8). Results show a 5% correction yielding a small increase from 2.1 to 2.2 overall in Equation (11).

Appendix F. Water Vapor Feedback and Radiation Energy Flux Breakdown with UHF

The consequences of a 13% increase in global warming due to urbanization introduces key issues both in making radiative warming estimates and how it affects atmospheric water vapor. When the global air temperature increases, it causes an expansion of atmospheric gases and allows the air to hold more water vapor before saturation. Thus, warmer air holds more water vapor. For example, from 1950 to 2019, a 0.95 °C rise occurred. The rule of thumb is the increase in water vapor goes as the temperature ratio, in this case

$$T_2/T_1 = 15.45 \ ^{\circ}C \ /14.5 \ ^{\circ}C = 1.0655$$
 (A13)

Here $T_1 = 14.5$ °C, the 1950 estimated average air temperature, and $T_2 = 15.45$ °C, the increase in 2021. We can refine this with the Clausius-Clapeyron relative humidity relationship

Water Vapor Factor =
$$Exp[-2.465 E 6/462\{1/T_1 - 1/T_2\}] = 1.0629$$
 (A14)

Here T_1 and T_2 are in degrees K, 2.465E6 J-kg⁻¹ is the latent heat of vaporization and 406 J-kg⁻¹ K⁻¹ (287) is the specific gas constant for water vapor. We note Equations (A13) and (A14) are in good agreement.

The average estimated composition of the atmosphere up to an altitude of 25 km with CO_2 is about 400 ppm and water vapor is about 25,000 ppm [78].

Using this water vapor value, and Equation (A14), the average will increase by 1573 ppm (=25,000 \times 1.0629–25,000) since the average air is warmer by the 0.95 °C temperature rise.

Then given our results that the UHF forcing is about 13%, this portion yields 204 ppm, leaving 1368 ppm due to the general warming of GHGs leaving a breakdown for the increase in average water vapor increase as

Water Vapor Increase (1950 to 2021) =
$$(204 \text{ ppm})_{\text{UHF}} + (1368 \text{ ppm})_{\text{CHCs}}$$
 (A15)

We have been using the radiation warming increase from 1950 to 2018 of 5.1 Wm⁻². If we extrapolated from IPCC/NOAA tables for the period between 1950 and 2019, yielding 2.38 W/m² estimated by Butler and Montzka [79]; see also Feinberg, [9] the estimates for change in the radiation energy flux ΔR , yields the following breakdown

$$\Delta R = \left(1.78Wm^{-2}\right)_{CO_2} + \left(0.6Wm^{-2}\right)_{other GHGs} + \left(2.72Wm^{-2}\right)_{H_2O} = 5.1Wm^{-2} \quad (A16)$$

The problem is, this IPCC estimate does not account for the UHF energy contribution that is estimated here as 0.66 Wm^{-2} (13% of 5.1 Wm^{-2}). This includes its feedback effect.

Without feedback it forcing is 0.307 Wm^{-2} . This is the UHF forcing (= $0.66 \text{ Wm}^{-2}/2.15$). We can consider the following proportionate correction reducing other by GHGs 25% and CO₂ by 75% of the UHI value. Then Equation (A16) changes to

$$\Delta RW = \begin{pmatrix} \left(1.55Wm^{-2}\right)_{CO_2} + \left(0.52Wm^{-2}\right)_{other GHGs} \\ + \left(0.31Wm^{-2}\right)_{UHF} \end{pmatrix}_{Forcing} + \left(2.72Wm^{-2}\right)_{Feedbk}$$
(A17)
= 5.1Wm^{-2}

Note that the 0.31 Wm^{-2} urbanization forcing represents 13% of the forcing (=0.31 $\text{Wm}^{-2}/(1.55 + 0.52 + 0.31) \text{Wm}^{-2}$). The attribution change is 13% less GHG and 13% added urbanization effect. Note this 13% ratio is maintained just as the full urbanization warming contribution with a feedback value of 0.66 Wm^{-2} (=2.15 × 0.31 Wm^{-2}) is also 13% of total warming feedback and forcing 5.1 Wm^{-2} flux.

We can assess our carbon dioxide correction estimate using a forcing relationship by Myhre et al. [80] (see also ASC [81]) with values of 311 ppm in 1950 and 415 ppm in 2019 for CO_2 . This indicates a radiation-forcing estimate of

$$1.54 Wm^{-2} = 5.35 Wm^{-2} Ln(415/311)$$
 (A18)

This is fairly close to our estimate for CO_2 forcing correction estimate of 1.55 Wm⁻² in Equation (A17), indicating some confidence in our estimate.

We can also use Equation (A18) for water vapor if we know its relative strengths compared to CO_2 . Kiehl et al. [82] estimated that water vapor was about 2.3 times stronger than CO_2 forcing on a clear sky day, but larger on cloudy days (about 33% larger). Consider the estimated increase in the average water vapor of 25,000 ppm of 1573 ppm in Equation (A15). This average radiation warming could be compared to an area having 20,000 ppm of water vapor. We can compare these two areas with different water vapor IR downwellings by using Equation (A18) in a similar way by applying a CO_2 relative strength factor of f = 3 to water vapor yielding a very rough estimate of,

$$5.35 \, \text{Wm}^{-2}\{(\text{Ln}[(25,000+1573\times f)/20,000)]\}) = 2.1 \, \text{Wm}^{-2} \tag{A19}$$

This is just a helpful example demonstrating the logarithmic saturation of water vapor that provides an idea of how a substantial increase of 1573 ppm compared to a 100 ppm increase of CO_2 , only leads to a radiative warming effect of 1.55 Wm⁻² estimate in Equation (A17). This suggests a possible average scenario. However, the purpose of the example is to help illustrate possible warming causality and also the difficulty of climate science in making such estimates.

Appendix G. Satellite Issues in Urbanization Assessments

Satellite measurements such as those reported by Guo et al., [19] and Ouyang et al. [21] conclude a negligible or even a reverse forcing albedo effect from urbanization. However, Ouyang et al. [21] quoted the area assessed as about 0.15% of the earth which also include farmlands. This is about 1.7 times smaller than Sun et al.'s [27] estimate for ISAs. This is likely partly due to the difference in accounting for ISA building sides. Ouyang et al. [21] suggest that effective radiative forcing (ERF) modeling may be a better tool than radiative forcing (RF) assessment, but provide assessments with simple RF measurements that ignore complex vertical building structure sides with solar heating of ISAs in UHIs. Satellite results make comparisons to rural areas. However, heat fluxes in cities are subjected to amplification effects and these need to be considered. Furthermore, in this study, LSCT assessments were an added metric that is part of our everyday experiences of solar heating of ISAs heat fluxes in city pavements, roofs, building sides, and on millions of miles of rural black asphalt roads. This was assessed relative to ambient conditions. In this study, we focus on ISAs, not all urbanization albedos which can include farmland for example.

To this point, Kiersten and Möller [18] note that "Remote sensing has several limitations including capturing emissions from vertical surfaces" such as the sides of building walls. This is a challenge for remote sensing because the equipment is designed to capture missions from horizontal surfaces such as streets, rooftops, and treetops [83]. The sensor captures data from a birds-eye point of view which is not always the most accurate representation of the UHI. The second challenge is that remotely sensed data represents radiation that has traveled through the atmosphere twice [83]. As the sensor receives the measurement, the wavelength has traveled from the sun to the earth and then reflected from the earth's surface back into the atmosphere to an orbiting or high flight sensor. To account for this error the data must be adjusted to include solar reflectance and temperature [83]. He et al. [20] found large albedo satellite discrepancies: "At latitudes, higher than 50°, the maximal difference in winter zonal albedo ranges from 0.1 to 0.4 among the nine satellite data sets". A study in optimum conditions with clear sky and reasonably flat terrain of the Atacama Desert by Cordero et al. [84] found: "Satellite-derived estimates for sites in the southernmost part of the Atacama Desert ($27-30^{\circ}$ S) exhibited negative relative bias errors (up to about -20%). These larger discrepancies can be attributed to the scattered vegetation in the region and the difference in MODIS and our instruments ... The relatively lower signal-to-noise ratio of our ground measurements ... the reflectance (and in turn the signal) was relatively low over the range 2105–2155 nm due to a water vapor absorption band ... These types of numerous challenges in Ouyang et al. [21] satellite approach are not well addressed in their paper. By comparison, Zhang et al. [1] provided a detailed history of ground-based measurement issues and their precautions.

The satellite measurements may not accurately assess hot spot areas that likely dominate warming issues because they raise the ISA average temperature value and may require higher resolution to estimate actual global averages. As mentioned a solar-heated rural road and one in a city can almost be equal in temperature within the ambient temperature differences which in comparative assessments values may be compromised. As well, assessing albedo can be difficult; ground measurements find very large albedo pavement variations seasonally, due to aging, and material type. Therefore, satellite results may have large variabilities that are not addressed.

Theobald et al. [56] argued that "remotely sensed imagery has limitations because it can require human interpretation to classify adequately and can miss development features that are obstructed by vegetation canopy or are small or narrow features".

Satellite results assessed by Ouyang et al. [56] did not recognize the results in the Zhang et al. [1] study. Furthermore, a full assessment of contact surface temperature observations worldwide may need very high resolutions to resolve all asphalt streets and roof temperatures. As well building sides are likely a key issue where satellite measurements are unable to accurately assess. Furthermore, UHIs amplify heat from the albedo effect which should be taken into account when making comparisons to neighboring rural areas. For comparisons, we discuss the relative natural reflectivity of land (~25%) in Section 3.3.1 and how this may be insightful. These are just some of the issues that may be problematic in remote sensing assessment and could be key reasons for their conflicts with Zhang et al.'s results and Table 7. We discuss this as many authors (IPCC) favor satellite albedo land cover assessments ignoring direct measurement ground-based studies that intrinsically include amplification effects, and it is apparent these create key discrepancies.

References

- 1. Zhang, P.; Ren, G.; Qin, Y.; Zhai, Y.; Zhai, T.; Tysa, S.K.; Xue, X.; Yang, G.; Sun, X. Urbanization effects on estimates of global trends in mean and extreme air temperature. *J. Clim.* **2021**, *34*, 1923–1945. [CrossRef]
- IPCC AR6. Regional Fact Sheet—Urban Areas. 2021. Available online: https://www.ipcc.ch/report/ar6/wg1/downloads/ factsheets/IPCC_AR6_WGI_Regional_Fact_Sheet_Urban_areas.pdf (accessed on 11 January 2023).
- 3. Feinberg, A. Climate change trends due to population growth: Feedback and CO₂ doubling temperature opposing rates. *Preprint* **2023**. *submitted*.
- Worldbank. Urban Development. 2020. Available online: https://www.worldbank.org/en/topic/urbandevelopment/overview# 1 (accessed on 4 December 2021).

- IEA (International Energy Agency). World Energy Outlook. 2018. Available online: https://iea.blob.core.windows.net/assets/77 ecf96c-5f4b-4d0d-9d93-d81b938217cb/World_Energy_Outlook_2018.pdf (accessed on 23 December 2022).
- 6. Yang, Q.; Huang, X.; Tang, Q. The footprint of urban heat island effect in 302 Chinese cities: Temporal trends and associated factors. *Sci. Total Environ.* **2019**, *665*, *652–662*. [CrossRef] [PubMed]
- Koerner, B.; Klopatek, J. Anthropogenic and natural CO₂ emission sources in an arid urban environment. *Environ. Pollut.* 2002, 116, S45–S51. [CrossRef]
- Bohnenstengel, S.; Davis, M.; Belcher, S. Impact of anthropogenic emissions on London's temperatures. *Q. J. R. Meteorol. Soc.* 2014, 140, 687–698. [CrossRef]
- 9. Feinberg, A. A Re-Radiation Model for the Earth's Energy Budget and the Albedo Advantage in Global Warming Mitigation. *Dyn. Atmos. Ocean.* **2022**, *97*, 101267. [CrossRef]
- 10. Block, A. Impacts of anthropogenic heat on regional climate patterns. Geophys. Res. Lett. 2004, 31, L12211. [CrossRef]
- 11. Makar, P.A.; Gravel, S.; Chirkov, V.; Strawbridge, K.B.; Froude, F.; Arnold, J.; Brook, J. Heat flux, urban properties, and regional weather. *Atmos. Environ.* **2006**, 40, 2750–2766. [CrossRef]
- 12. Narumi, D.; Kondo, A.; Shimod, Y. Effects of anthropogenic heat release upon the urban climate in a Japanese megacity. *Environ. Res.* **2009**, *109*, 421–431. [CrossRef]
- 13. Feng, J.-M.; Wang, Y.-L.; Ma, Z.-G.; Liu, Y.-H. Simulating the Regional Impacts of Urbanization and Anthropogenic Heat Release on Climate across China. J. Clim. 2012, 25, 7187–7203. [CrossRef]
- 14. Yang, W.; Luan, Y.; Liu, X.; Yu, X.; Miao, L.; Cui, X. A new global anthropogenic heat estimation based on high-resolution nighttime light data. *Sci. Data* **2017**, *4*, 170116. [CrossRef] [PubMed]
- 15. Flanner, M.G. Integrating anthropogenic heat flux with global climate models. Geophys. Res. Lett. 2009, 36, L02801. [CrossRef]
- 16. Zhao, L.; Lee, X.; Smith, R.; Oleson, K. Strong, contributions of local background climate to urban heat islands. *Nature* **2014**, *511*, 216–219. [CrossRef] [PubMed]
- 17. Feinberg, A. Urban Heat Island High Water-Vapor Feedback Estimates and Heatwave Issues: A Temperature Difference Approach to Feedback Assessments. *Science* 2022, *4*, 44. [CrossRef]
- Kiersten, L.; Möller, G. Urban Heat Island Effect: Measuring the Urban Heat Island Effect. In *Principles of Sustainability*; University of Idaho: Moscow, ID, USA, 2022; Chapter 4. Available online: https://www.webpages.uidaho.edu/sustainability/ch04-p05d. https://www.webpages.uidaho.edu/sustainability/ch04-p05d. https://www.webpages.uidaho.edu/sustainability/ch04-p05d. https://www.webpages.uidaho.edu/sustainability/ch04-p05d. https://www.webpages.uidaho.edu/sustainability/ch04-p05d. https://www.webpages.uidaho.edu/sustainability. https://www.webpages.uidaho.edu
- Guo, T.; He, T.; Liang, S.; Roujean, J.; Zhou, Y.; Huang, X. Multi-decadal analysis of high-resolution albedo changes induced by urbanization over contrasted Chinese cities based on Landsat data. *Remote Sens. Environ.* 2022, 269, 112832. Available online: http://www.sciencedirect.com/science/article/pii/S0034425721005526 (accessed on 22 November 2022). [CrossRef]
- 20. He, T.; Liang, S.; Song, D.X. Analysis of global land surface albedo climatology and spatial-temporal variation during 1981–2010 from multiple satellite products. *J. Geophys. Res. Atmos.* **2014**, *119*, 10281–10298. [CrossRef]
- 21. Ouyang, Z.; Sciusco, P.; Jiao, T.; Feron, S.; Lei, C.; Li, F.; John, R.; Fan, P.; Li, X.; William, C.; et al. Albedo changes caused by future urbanization contribute to global warming. *Nat. Commun.* **2022**, *13*, 3800. [CrossRef]
- 22. Ritchie, H.; Roser, M. Energy Production and Consumption, Our World in Data. 2022. Available online: https://ourworldindata. org/energy-production-consumption (accessed on 29 December 2022).
- Liu, R.; Su, H.; Liou, K.; Jiang, J.; Gu, Y.; Liu, S.; Shiu, C. An Assessment of Tropospheric Water Vapor Feedback Using Radiative Kernels. JGR Atmos. 2018, 123, 1499–1509. [CrossRef]
- 24. Hartmann, D.L.; Klein Tank, A.M.G.; Rusticucci, M.; Alexander, L.V.; Brönnimann, S.; Charabi, Y.; Dentener, F.J.; Dlugokencky, E.J.; Easterling, D.R.; Kaplan, A.; et al. Observations: Atmosphere and surface. In *Climate change* (2013) the Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change; Cambridge University Press: Cambridge, UK; New York, NY, USA, 2013.
- 25. Feinberg, A. Urban heat island amplification estimates on global warming using an albedo model. *SN Appl. Sci.* **2020**, *2*, 2178. [CrossRef]
- Cherlet, M.; Hutchinson, C.; Reynolds, J.; Hill, J.; Sommer, S.; von Maltitz, G. (Eds.) *Distribution of the world's Big Cities, World Atlas of Desertification*; Publication Office of the European Union: Luxembourg, 2018. Available online: https://wad.jrc.ec.europa.eu/aridityurban (accessed on 20 November 2022).
- 27. Sun, Z.; Du, W.; Jiang, H.; Weng, Q.; Guo, H.; Han, Y.; Xing, Q.; Ma, Y. Global 10-m impervious surface area mapping: A big earth data based extraction and updating approach. *Int. J. Appl. Earth Obs. Geoinf.* **2022**, *109*, 102800. [CrossRef]
- Huang, X.; Yang, J.; Wang, W.; Liu, Z. Mapping 10 m global impervious surface area (GISA-10m) using multi-source geospatial data. *Earth Syst. Sci. Data* 2022, 14, 3649–3672. [CrossRef]
- Dessler, A.; Zhang, Z.; Yang, P. Water-vapor climate feedback inferred from climate fluctuations, 2003–2008. *Geophys. Res. Lett.* 2008, 35, 20. [CrossRef]
- 30. Gordon, N.; Jonko, A.; Forster, P.; Shell, K. An observationally based constraint on the water-vapor feedback. *J. Geophys. Res.* 2013, 118, 12435–12443. [CrossRef]
- 31. Pigeon, G.; Legain, D.; Durand, P.; Masson, V. Anthropogenic heat release in an old European agglomeration (Toulouse, France). *Int. J. Climatol.* **2007**, *27*, 1969–1981. [CrossRef]
- Chen, B.; Dong, L.; Shi, G.; Lli, L.; Chen, L. Anthropogenic Heat Release: Estimation of Global Distribution and Possible Climate Effect. J. Meteorol. Soc. Japan. Ser. II 2014, 92, 157–165. [CrossRef]

- 33. Zhou, D.; Zhao, S.; Zhang, L.; Sun, G.; Liu, Y. The footprint of urban heat island effect in China. Sci. Rep. 2015, 5, 11160. [CrossRef]
- 34. Fan, Y.; Li, Y.; Bejan, A.; Wang, Y.; Yang, X. Horizontal extent of the urban heat dome flow. *Sci. Rep.* **2017**, *7*, 11681. [CrossRef]
- Wang, P.; Yang, Y.; Tang, J.; Leung, L.; Liao, H. Intensified Humid Heat Events Under Global Warming. *Geophys. Res. Lett.* 2021, 48, e2020GL091462. [CrossRef]
- Schär, C.; Vidale, P.L.; Lüthi, D.; Frei, C.; Häberli, C.; Liniger, M.A.; Appenzeller, C. The role of increasing temperature variability in European summer heatwaves. *Nature* 2004, 427, 332–336. [CrossRef]
- 37. Li, D.; Bou-Zeid, E. Synergistic interactions between urban Heat Islands and heat waves: The impact in cities is larger than the sum of its parts. *J. Appl. Meteorol. Climatol.* **2013**, *52*, 2051–2064. [CrossRef]
- 38. Russo, S.; Sillmann, J.; Sterl, A. Humid heat waves at different warming levels. Sci. Rep. 2017, 7, 7477. [CrossRef] [PubMed]
- Yang, X.; Ruby Leung, L.; Zhao, N.; Zhao, C.; Qian, Y.; Hu, K.; Liu, X.; Chen, B. Contribution of urbanization to the increase of extreme heat events in an urban agglomeration in east China. *Geophys. Res. Lett.* 2017, 44, 6940–6950. [CrossRef]
- Liao, W.; Liu, X.; Li, D.; Luo, M.; Wang, D.; Wang, S.; Baldwin, J.; Lin, L.; Li, X.; Feng, K.; et al. Stronger Contributions of Urbanization to Heat Wave Trends in Wet Climates. *Geophys. Res. Lett. Am. Geophys. Union (AGU)* 2018, 45, 11310–11317. [CrossRef]
- 41. Li, C.; Sun, Y.; Zwiers, F.; Wang, D.; Zhang, X.; Chen, G.; Wu, H. Rapid Warming in Summer Wet Bulb Globe Temperature in China with Human-Induced Climate Change. *J. Clim.* **2020**, *33*, 5697–5711. [CrossRef]
- 42. Kong, D.; Gu, X.; Li, J.; Ren, G.; Liu, J. Contributions of Global Warming and Urbanization to the Intensification of Human-Perceived Heatwaves Over China. *JPG Atmos.* **2020**, *125*, e2019JD032175. [CrossRef]
- 43. Zou, Z.; Yan, C.; Yu, L.; Jiang, X.; Ding, J.; Qin, L.; Wang, B.; Qiu, G. Impacts of land use/land cover types on interactions between urban heat island effects and heat waves. *Build. Environ.* **2021**, 204, 108138. [CrossRef]
- 44. Dewan, A.; Kiselev, G.; Botje, D.; Mahmud, G.; Bhuian, H.; Hassan, Q. Surface urban heat island intensity in five major cities of Bangladesh: Patterns, drivers and trends. *Sustain. Cities Soc.* **2021**, *71*, 102926. [CrossRef]
- 45. Zhang, S.; Ren, G.; Ren, Y.; Tysa, S. Linkage of extreme temperature change with atmospheric and locally anthropogenic factors in China mainland. *Atmos. Res.* **2022**, 277, 106307. [CrossRef]
- 46. Yu, H.; Liu, S.; Dickinson, R. Radiative effects of aerosols on the evolution of the atmospheric boundary layer. *J. Geophys. Res.-Atmos.* **2002**, *107*, AAC 3–1–AAC 3–14. [CrossRef]
- 47. Lai, L. The influence of urban heat island phenomenon on PM concentration: An observation study during the summer half-year in metropolitan Taipei, Taiwan. *Theor. Appl. Climatol.* **2018**, *131*, 227–243. [CrossRef]
- Potter, C.; Alexander, O. Impacts of the San Francisco Bay Area shelter-in-place during the COVID-19 pandemic on urban heat fluxes. Urban Clim. 2021, 37, 100828. [CrossRef] [PubMed]
- Cao, C.; Lee, X.; Liu, S.; Schultz, N.; Xiao, W.; Zhang, M.; Zhao, L. Urban heat islands in China enhanced by haze pollution. *Nat. Commun.* 2016, 7, 12509. [CrossRef] [PubMed]
- Feinberg, A. A Molecular Level Study Indicates the Capability of CO₂ Warming Amplification in UHIs Can Significantly Impact Microclimate Warming to the Global Level. *Preprint*. 2021. Available online: https://www.researchgate.net/publication/35 2466630_A_Molecular_Level_Study_Indicates_the_Capability_of_CO2_Warming_Amplification_in_UHIs_can_Significantly_ Impact_Microclimate_Warming_to_the_Global_Level (accessed on 29 November 2021).
- 51. NASA Global Climate Change. 2022. Available online: https://climate.nasa.gov/vital-signs/global-temperature/ (accessed on 25 November 2022).
- 52. Chaisson, E.J. Long-term global heating from energy use. *Eos Trans. AGU* 2008, 89, 253. [CrossRef]
- 53. Feinberg, A. Solar Geoengineering Modeling and Applications for Mitigating Global Warming: Assessing Key Parameters and the Urban Heat Island Influence. *Front. Clim.* **2022**, *4*, 870071. [CrossRef]
- 54. Liu, Z.; He, C.; Zhou, Y.; Wu, J. How much of the world's land has been urbanized, really? A hierarchical framework for avoiding confusion. *Landsc. Ecol.* **2014**, *29*, 763–771. [CrossRef]
- 55. Zhao, M.; Cheng, C.; Zhou, Y.; Li, X.; Shen, S.; Song, C. A global dataset of annual urban extents (1992–2020) from harmonized nighttime lights. *Earth Syst. Sci. Data* 2022, *14*, 517–534. [CrossRef]
- 56. Theobald, D.; Kennedy, C.; Chen, B.; Oakleaf, J.; Baruch-Mordo, S.; Kiesecker, J. Earth transformed: Detailed mapping of global human modification from 1990 to 2017. *Earth Syst. Sci. Data* 2020, 12, 1953–1972. [CrossRef]
- Liu, Z.; Lai, J.; Zhan, W.; Bechtel, B.; Voogt, J.; Quan, J.; Hu, L.; Fu, P.; Huang, F.; Li, L.; et al. Urban Heat Islands Significantly Reduced by COVID-19 Lockdown. *Geophys. Res. Lett.* 2022, 49, e2021GL096842. [CrossRef]
- Climate Central. Hot Zones: Urban Heat Islands, Research brief by Climate Central. 2021. Available online: https://assets. ctfassets.net/cxgxgstp8r5d/1XZZjkLYwtcmKL5k3wEinl/5f8c9b5b2d8dd56e1bda7f51278fc3d2/2021_UHI_Report.pdf (accessed on 20 November 2022).
- Rasmussen, C.; Lee, J.; Wang, A. NASA's ECOSTRESS Sees Las Vegas Streets Turn Up the Heat, Jet Propulsion Lab. 2022. Available online: https://www.nasa.gov/feature/jpl/nasa-s-ecostress-sees-las-vegas-streets-turn-up-the-heat (accessed on 25 November 2022).
- 60. Kowal-Vern, A.; Matthews, M.; Richey, K.; Ruiz, K.; Peck, M.; Jain, A.; Foster, K. "Streets of Fire" revisited: Contact burns. *Burn. Trauma* **2019**, *7*, s41038-019-0169-9. [CrossRef]
- 61. Gewirtz, E. How Hot Is Too Hot for a Dog's Paws?, American Kennel Club. 2021. Available online: https://www.akc.org/expertadvice/health/heatstroke-in-dogs/ (accessed on 29 November 2021).

- 62. Morabito, M.; Crisci, A.; Georgiadis, T.; Orlandini, S.; Munafò, M.; Congedo, L.; Rota, P.; Zazzi, M. Urban Imperviousness Effects on Summer Surface Temperatures Nearby Residential Buildings in Different Urban Zones of Parma. *Remote Sens.* **2018**, *10*, 26. [CrossRef]
- 63. Mentaschi, L.; Duveiller, G.; Zulian, G.; Corbane, C.; Pesaresi, M.; Maes, J.; Stocchino, A.; Feyen, L. Global long-term mapping of surface temperature shows intensified intra-city urban heat island extremes. *Glob. Environ. Chang.* 2022, 72, 102441. [CrossRef]
- 64. Knox, P. How Hot Does Pavement Get in Summer? *Climate and Agriculture in the Southeast*. 2022. Available online: https://site.extension.uga.edu/climate/2022/05/how-hot-does-pavement-get/ (accessed on 25 November 2022).
- 65. Feinberg, A. Department of Transportation Asphalt Roads Causing Heat & Heatwaves in Cities, YouTube presentation. 2022. Available online: https://www.youtube.com/watch?v=SXEm3XN8cr8 (accessed on 25 November 2022).
- 66. Sugawara, H.; Takamura, T. Surface albedo in cities (0.12): Case study in Sapporo and Tokyo, Japan. *Bound Layer Meteorol.* **2014**, 153, 539–553. [CrossRef]
- 67. Trlica, A.; Hutyra, L.; Schaaf, C.; Erb, A.; Wang, J. Albedo, Land Cover, and Daytime Surface Temperature Variation Across an Urbanized Landscape. *Earth's Future* **2017**, *5*, 1084–1101. [CrossRef]
- MIT Study on Roads; AzariJafari, H.; Kirchain, R.; Gregory, J. Mitigating Climate Change with Reflective Pavements. *CSHub Top. Summ.* 2020, 2020. Available online: https://cshub.mit.edu/sites/default/files/images/Albedo%201113_0.pdf (accessed on 25 November 2022).
- 69. Akbari, H.; Matthews, D.; Seto, D. The long-term effect of increasing the albedo of urban areas. *Environ. Res. Lett.* **2012**, *7*, 024004. [CrossRef]
- EPA. Reducing Urban Heat Islands: Compendium of Strategies. 2014. Available online: https://www.epa.gov/sites/default/ files/2014-08/documents/coolpavescompendium_ch5.pdf (accessed on 12 March 2023).
- 71. EPA. Study Cambridge Systematics. *Cool Pavement Report, Heat Island Reduction Initiative*. 2005. Available online: https://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.648.3147&rep=rep1&type=pd (accessed on 12 March 2023).
- Octopus Energy. 2023. Available online: https://octopus.energy/blog/when-will-fossil-fuels-run-out/#:~:text=If%20we%20 keep%20burning%20fossil,will%20be%20depleted%20by%202060 (accessed on 12 March 2023).
- Widory, D.; Javoy, M. The carbon isotope composition of atmospheric CO2 in Paris. *Earth Planet. Sci. Lett.* 2003, 215, 289–298. [CrossRef]
- 74. Gratani, L.; Varone, L. Daily and seasonal variation of CO₂ in the city of Rome in relationship with the traffic volume. *Atmos. Environ.* **2005**, *39*, 2619–2624. [CrossRef]
- 75. George, K.; Ziska, L.H.; Bunce, J.A.; Quebedeaux, B. Elevated atmospheric CO₂ concentration and temperature across an urban–rural transect. *Atmos. Environ.* **2007**, *41*, 7654–7665. [CrossRef]
- Velasco, E.; Pressley, S.; Allwine, E.; Westberg, H.; Lamb, B. Measurements of CO₂ fluxes from the Mexico City urban landscape. *Atmos. Environ.* 2005, 39, 7433–7446. [CrossRef]
- 77. NOAA Climat Ratios. 2021. Available online: https://gml.noaa.gov/outreach/lesson_plans/Climate%20Ratios.pdf (accessed on 16 December 2022).
- Ossfoundation. 2023. Available online: https://ossfoundation.us/projects/environment/global-warming/atmosphericcomposition (accessed on 23 March 2023).
- 79. Butler, J.; Montzka, S. The NOAA Annual Greenhouse Gas Index, Earth System Researh Lab. Global Monitoring Laboratory. 2020. Available online: https://www.esrl.noaa.gov/gmd/aggi/aggi.html (accessed on 23 March 2023).
- 80. Myhre, G.; Highwood, E.J.; Shine, K.P.; Stordal, F. New estimates of radiative forcing due to well mixed greenhouse gases. *Geophys. Res. Lett.* **1998**, *25*, 2715–2718. [CrossRef]
- ASC. 2023. Available online: https://www.acs.org/climatescience/atmosphericwarming/climatsensitivity.html (accessed on 23 March 2023).
- 82. Kiehl, J.T.; Trenberth, K.E. Earth's Annual Global Mean Energy Budget. Bull. Amer. Meteor. Soc. 1997, 78, 197–208. [CrossRef]
- 83. Natural Resources Canada, Fundamental of Remote Sensing. Available online: https://natural-resources.canada.ca/sites/www.nrcan.gc.ca/files/earthsciences/pdf/resource/tutor/fundam/pdf/fundamentals_e.pdf (accessed on 23 March 2023).
- Cordero, R.R.; Feron, S.; Sepúlveda, E.; Damiani, A.; Carrera, J.M.; Jorquera, J.; Alfonso, J.A.; Fuenzalida, R.; Rivas, M.; MacDonell, S.; et al. Evaluation of MODIS-derived estimates of the albedo over the Atacama Desert using ground-based spectral measurements. *Sci. Rep.* 2021, *11*, 19822. [CrossRef] [PubMed]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.